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GENERAL　INTRODUCTION

1　Fluidization

　　　When価d，　i．e．　gas　or　liquid，　is　supplied　into　a　bed　of　solid　par七cles　supported　by　a

horizonteil　porous，　perforated　or　gauze　plate　in　a　venical　column，　all　the　panicles　experience

an　upward　drag　force　exerted　by　th　e　fl　u　id　stream　as　well　as　the　gravitational　and　buoyant

forces．　Th　e　d　rag　fo　rce　i　ncreases　with　the　fi　u　id　flow　rate　leavi　ng　the　latter　two　forces

unchanged　and　the　increase　is　pra（：；tically　observed　as　an　increase　in廿1e　pressure　drop

through　the　bed．　When　the　pressure　drop　becomes　sufficient　tb　support　the　weight　of　the

bed　in　the　fl　uid　per　u　n　it　crosssecljon，　the　padicles　are　released　from　gravitation　and　are

sa’п@to　be　thctp’　ientiy　fiuidi　ed．　At　th　ls　condin’on，　the　bed　gains　a　fl　u　idA　ike　prope　nies　and

hence　is　called　as　a　fluidined　bed．　Terminologicaliy，　a　bed　fluidized　by　gas　is　referred　to　a・

gas－fiuidized　bed　while廿lat　fluidized　by　liquid　as　a〃ヒ7uわし伽たYized　bed．　　Fur廿1er，　a　bed

fiuidized　by　co－current　flow　of　gas　and　liquid　is　called　as　a　gasViquたPso化／fiuたtLzed　or　tわrθe－

phase　fiuidized　bed．　Such　a　fluid　operation　to　fluidtze　a　bed　of　panicles　is　genera”y　catled

月1．iた／ization．

　　　So　far，刊uidization　has　widely　been　tm’lized　as　one　of廿1e　promising　t㏄hniques　to　bhng

solid　partr’cles　and　fluid　into　continuous　contact　i　n　a　reactor．　lt　offers　a　good　oontact　of

them　promoting　heat　and　mass　transfers　between　them．　l　n　addin’on，　oontinuous　feed　and

discharge　of　solid　panicles　can　easily　be　achieved　in　a　fiuidized　bed．　From　these

advantages，　fiuidization　has　been　extensively　adopted　in　various　industn’al　processes　for

catatytic　as　well　as　non－catedytic　gas－solid　reactl’ons，　cetict　nati　on，　oombustion，　adsorption，

drying　and　so　on．　When　solid　parts’des　such　as　catalyst　and　ooal　are　oontinuously　fed　into

and　discharged　from　a　fluidized　bed　reactor　having　a　fi　nite　volume，　the　parddes　in　the

reactor　can　generally　be　assumed　to　be　perfeody　mixed．　This　is　because　rate　of　solid

mixing　is　usually　much　higher　than　those　of　gas－solid　reaction　in　the　above　mentioned

prcx　esses．　This　rapid　mixing　sometimes　makes　the　reacb’on　operation　inefficient　due　to　a

vvide　disbibution　of　solid　residence　time　and　oonsequentiy　a　wide　distribution　in　the　extent　of

reaction．　Circulation　of　panides　elutriated　from　a　reactor　at　high　gas　velocin’es　back　to　the

reactor　h　as　offered　a　practicat　solution　to　narrow　the　vvidely　distributed　solid　residence　time．

The　ciroulating　system　is　now　termed　as　a　circulating　fluidized　bed　（CFB）　system．　The

matn　components　of　CFB　system　are　generally　a　riser，　where　the　solid　particles　are

　　　　　　　　　　　　　　　　　　　　　　　　　　　　－1一
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fluidized　and　entratned　upward　at　fiuid　velociv’es　m　uch　higher　than　the　term　inal　velocity　of　a

single　particle，　and　a　downoomer，　where　th　e　elutriated　partides　from　the　riser　are　stored

and　recycled　to　th　e　riser．　Though　industrial　applications　of　C　FB　started　in　1　940s　and　were

rapidly　expanded　in　the　last　two　decades　stimulating　fundamentat　research，　our　knowledge

of　the　motion　of　panides　fluidized　at　such　high　fluid　velociti‘es　in　the　riser　has　st）’ll　been　far

from　suthcient　for　reliable　design　of　the　system．　ln　parti’cular，　some　inconsistent

macroscopic　fiow　pattems　have　been　reported　by　d　nterent　i　nvest）’№≠狽盾窒刀@who　wotked　with

dnterent　experimentat　systems　under　different　operating　condnions．　lt　is　clear　from

literatu　re　that　the　fiow　pattem　strongly　depends　on　the　scale　and　oonfigurations　of　CFB

system　as　well　as　on　the　operati　ng　van’ables．　Th　us，　the　pu　rpose　of　the　present　study　is　to

elua’date　the　solid　flow　pattems　as　a　fu　nCtjon　of廿1e　above　vahaiCAes．

2　Thesis　Summary

　　　This　thesis　oonsists　of　six　chapters　which　are　summarized　as　follows：

　　　ln　Chapte　r　1，　a　critical　review　is　given　for　existing　i　nvesti’gations　on　hyd　rodynamic．

characterist）’cs　in　CFB　risers　to　make　clear　the　background　and　pu　rpose　of　the　present

study．

　　　Chapter　2　deals　wtth　effects　of　the　riser　height　and廿1e　downoomer　solid　inventory　in

a　cirou　lating　gas－fiuidized　bed　（CGFB）　system　on　the　axiEal　bed　dens’rty　（ABD）　distributi’on　in

the　riser　and　on　the　stability　of　solid　circulation．　Experiments　were　carn’ed　out　using　two

dnterent－height　risers　（5．5　m　and　10　m）　having　an　inner　diameter　of　O．1　m．　Para’cles　were

fed　from　the　downoomer　bottom　to　the　riser　bottom　through　a　screw　feeder　at　van’ous

vatues　of　the　solid　mass　flux　regulated　independentiy　of　the　gas　velocity　in　the　riser　and　of

the　downoomer　solid　inventory．　ms　shown廿1at廿ie　ABD　distrl’bution　strongly　depends　on

the　downoomer　solid　inventory　when　the　solid　flux　takes　a　crit）’cal　value　at　a　g’Nen　gas

velocity．　Fu曲emore，　at　meα浦。創soiid　mass　flux　and　gas　veloc耐，　a　signMcant　vari◎ation

of　the　static　pressure　with　time　has　been　detected　at廿1e　bottom　of　the　downoomer，　which　is

atways　accompan　ied　by　that　of　the　A　B　D　d　istn’buti’on．

　　　Hence，　Chapter　3　describes　variations　of　the　static　pressure　at　the　downcomer

bottom　with　time，　wh　ich　were　obser〈／ed　at　the　abovementioned　criv’cal　condta’ons．　Their

correspondence　is　exam　ined　from　simuitaneous　observations　of　transhional　changes　in　the

ABD　distribution，　static　pressure　at　the　riser　bottom　and　circulating　solid　mass　flux　in　the

CG　FB　system．　lt　is　demonstrated　th　at　the　ci　rculati　ng　system　expe　riences　a　long　unsteady

State　after　all　operating　variables　are　fixed．　During　the　u　nsteady　state，　both　AB　D

disbibut’on　and　solid　mass　flux　are　shown　to　vary　with　the　difference　between　pressures　at

　　　　　　　　　　　　　　　　　　　　　　　　　　　－2一
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the　riser　and　the　downcomer　bottoms．　When　the　pressure　difference　between　the　riser

and　downoomer　bottoms　becomes　sedied，　a　steady　solid　m　ass　fl　ux　is　i　ndicated　to　be

attajnable．　Even　at廿1e　steady　operating　cond市ons，　an　apPredable　change　in　ABD

d’stn’bub’on　wth　time　is　observed，　which　is　attn’butable　to　the　changes　with　the　static

pressure　at　the　downoomer　bottom．

　　　As　described　in　the　preceding　chapters，　solid　movements　in　the　CG　FB　systems　are

affected　not　oniy　by　the　operati　ng　van’ables　but　by　the　system　configuration．　l　n　partiou　lar，

the　ABD　distributs’on　exhibits　a　very　complicated　response　to　the　operational　van’ables，

forrning　various　sizes　of　solid　dusters　in　the　lower　part　of　the　riser．　To　simpltfy　such　a　solid

movement，　a　circu　lating　liquid．fluidized　bed　（CLFB）　system　was　applied．　The　systematic

observations　were　perform　ed　for　the　static　pressure　disbibution　in　the　riser　at　fairiy　high

relative　velocin’es　of　liquid　to　solid　and　the　results　are　summarized　in　Chapter　4．　Based　on

systematic　measurements　of　the　static　pressures　along　the　height　of　a　riser　（4．0　m　high　and

O．049　m　i．d．）　at　various　supedia’at　liquid　velocity　and　circulating　solid　mass　flux，　effects　of　the

operating　variables　on　variations　in　the　bed　voidage　and　the　liquid－solid　slip　velocity　were・

examined．　An　image　anatysis　of　photographs　taken　through　the　riser　wall　was　atso

canied　out　to　quantify　fl　u　idization　states　of　th　e　solid　padides　i　n　term　s　of　the　power　spectrel

dens仕ky　distribution　and廿1e　fra（ねl　dimension．

　　　lt　is　demonstrated　that　the　panicles　fl　uidtzed　at　velocities　m　uch　h　igher　than　the

terminat　velocity　of　single　partjde　exhibit　to　an　aggregative　fluidization　state　where　the　loced

bed　voidage　and　the　panicle　movements　vigorously　fluctuate　with　time．　Thus，　in　Chapter

5，　the　complex　fluctuations　in　the　local　bed　voidage　and　the　heat　transfer　were

simutaneously　measured　using　an　optical　fiber　probe　and　a　heat　transfer　probe　and　the

time　series　data　was　analyzed　on　the　basis　of　a　deterministic　chaos　theory．　As　a　resutt，

the　complex仕y　of　the　voidage　and　heat　transfer　is　shown　to　be　charaderized　by廿1e

correlation　dimension　and　Kolmogorov　entropy　for　the　attractor．

　　　Chapter　6　gives　generat　oondusions　from　the　present　study　with　some　indications　for

further　work　needed　i　n　futu　re．
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Chapter　t

REVIEW　OF　FUNDAMENTAL　STUDIES　ON　HYDRODYNAMICS
　　　　　　　　　　　IN　CIRCULATING　FLUIDIZED　BEDS

1．1　Gas　Fluidization

1．1．1　Fluidization　Phenomena

　　　　The　flow　state　of　gas－fluidized　bed　in　a　venical　column　considerably　varies　with　the

superficiat　gas　velocity，　ugp，　as　depicted　in　Fig．1．1．　When　ugo　is　fairiy　low，　gas　simply

perGolates廿irough　the　interstice　among　stationary　patU’cles　and　the　particle　bed　is　called　a

fixec／　bed．　At　a　certain　value　of　upo　wh　ich　gives　the　upward　drag　force　equat　to　the

downward　gravitational　force　on　the　panicles，　the　particles　are　suspended　dynamically　in

the　gas　flow．　The　velocity　is　referred　as　the　minimum　or　in（ti）Aθηf去たfiraがbηveわGめ4　uml，

and　the　bed　at　umf　is　referred　as　the　minimum　or　incioienめ／fiuた〃ized　bθd．　　For　beds

consis廿ng　of　fine　particles　which　belong　to　the　scトcailed　Geldart　A　group（Geldart；1973），

they　expand　homogeneously　without　forming　bubbles　（particulate　f／uldization　or

んomqgθηθoαS衡た／izlヨtion）when　ug。　iS　inCreased　Slightly　Over　umf．　W辻h　a　fu曲er　increaSe

of　ug3，　bubbles　form　above　the　distributor　and　rise　through　the　bed．　The　veloc’rty　at　which

th　e　bubbles　start　to　appear　i　n　the　bed　is　referred　as　the　minimum　bubb／ing　ve／Ocity，　umb．

For　the　Geldart　B　and　D　group　particles，　the　particulate　fluidization　state　is　seldom

observed，　i．e．，　umb　usuaily　coincides　with　umf．　The　size　and　rising　velocity　of　bubbles

depend　on　operating　van’ables　and　particle　properties　as　investigated　by　many　workers．　A

bed　is　caHed　a　sZσ99加9加た／ized　bθct　when　bubbles　grow　in　size　to　the　bed　diameter．　In

these　bubbling　and　slugging　fluidized　beds，　solid　para’cles　and　intersthial　gas　form　a

continuous　emulsion　phase　while　gas　forms　a　dispersed　phase　of　bubbles　and　slugs．　A

further　increaSe　of　uge　brings　about　a　transition　from　the　bubbling　of　slugging　state　to　a

tunbu’ent衡た／i：zaton　state．　ln　the　latter　state，　the　boundary　of　bubble　or　slug　becomes

indistinct　and　the　emulsion　phase　is　seen　to　be　no　more　continuous　phase．　Furthermore，

the　padicles　irregulariy　move　upward　and　downward　in　a　highly　turbulent　manner　and

vigorous　pressure　fluctuations　are　observed　as　a　consequence　of　turbulent　interaou’on

between　gas　and　particles．

一4一
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　　　　At　ugp　higher　than　the　particle　terminal　velocity，　ut，　most　of　the　particles　are　blown

away　from　the　column　and　panicie　fluidization　of　the　panide　bed　in　the　column　becomes

impossible．　However，　if　the　particles　elutriated　from　the　oolumn　are　steadily　recyded　to　its

bottom，　they　rise　through　the　column　in　a　form　of　aggregative　clusters．　Here，　the　column

is　cadled　as　a　riser　and　a　system　consisting　of　solid　circulation　devices　such　as　a　cyclone，　a

downoomer　and　a　solid　feeder　together　with　the　riser　is　referred　as　a　circu／ating　gas－

fiuidized　bed　（CGFB）　system．　Yerushalmi　and　Cancurt　（Yerushalmi　and　Cancurt；　1979）

first　made　a　systematic　measurement　of　the　bed　yoidage　and　pressure　fluctuations　in　the

riser　of　a　CGFB　system．　Their　result　is　shown　in　Fig．t．2．　Here，　the　gas　velocity　giving

the　maximum　amplitude　in　the　pressure　fluctuation　is　defined　as　an　onset　veloc’rty　of

transi廿on　from廿1e　bubbling　fluidization　state　to　the　turbuient　fluidization　state，　Also，廿1e

offset　velocity　of　the　trans出on　is　defined　as廿1e　velocity　at　which　the　amp肚ude　falls　off．

Following　the　above　findings，　several　investigators　examined　characteristics　of　the　turbulent

fluidization　state　（Canada　et　al；　1978，　Horio；　1986，　Hashimoto　et　al；　1988，　Nakajima　et　al；

199q　Bi　et　al；1993）．　They　indicated　that廿1e　particle　movement　in　the　state　depends　not

only　on　uge　but　also　on　the　circulating　solid　flux，　Gs．　l　n　addition，　an　another　transition　takes

place　at　a　very　high　uge　from　the　turbulent　fluidization　to　a　pneumatic　transport　state，　where

panictes　are　dispersed　homogeneously　gMng　a　dilute　beci　with　pressu　re　fiuctuations　much

smatler　than　those　in　the　former　state．

1．1．2　Stability　of　Fluidiration

　　　　Wilhelm　and　Kwauk　（1948）　conducted　a　systematic　measurement　of　the　bed

pressure　drop　at　various　fluid　velocities　in　both　gas一　and　liquid－fluidized　beds　and

suggested　that　the　transition　between　the　particulate　and　aggregative　fluidization　states

takes　place　depending　on　Froude　number　（　＝　ugo2g’i　dp’i）．　Jackson　（1963）　first　made　a

theoretical　approach　to　explanation　of　the　transit’ton　based　on　a　perturbation　analysis．　The

analysis　was　performed　by　exerting　a　smail　local　disturbance　in　the　panicle　concentration

onto　an　incipiently　fluidized　bed　and　the　stab“ity　was　examined　in　terms　of　the　rate　of

upward　and　downward　propagation　of　the　distutoance　through　the　particle　bed．

Comparing　the　rates　of　proPagation　in　gas－fiuidized　beds　with　those　in　liquicFfluidized’　beds，

he　indicated　that　liquicFfluidized　beds　in　general　are　much　more　stable　than　gas－fluidized

beds　so　that　particle　beds　usually　exhibit　particulate　fluidization．　He　also　suggested　that

even　the　liquid－fluidized　beds　possibly　exhibit　an　aggregative　fluidization　state　which　is

similar　to廿1e伽idization　charaCteristics　in　gas－fluidized　beds　when　heavy　particles　such　as

lead　shot　or　copper　particles　are　employed　as　a　bed　material．　Recently，　Gibilaro　et　at
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（1986）sem’卜empirically　derived　a　trans｝廿on　crkeria　for　the　fluidization　states　based　on　a

dynamic　wave廿）eory（Wallis；1969）and　showed廿）at　beds　conslsting　of　fine　spherical

particles　always　exhibit　the　parb’culate　fluidization．

1．2　Circulating（tas－Fluidized　Bed（CGFB）Systems

1．2．1　Featu　res

　　　　CGFB　system　was　first　applied　to廿）e　upflow－type　fluid　catalytic　cracking　process

（Kunii　and　Levenspiel；　1991）　shown　in　Fig．1．3．　The　system　was　suocessfully　introducExS

as　an　ethcient　petroleum　refinery　technology　on　commercial　scale　to　overcome　the

shortage　of　gasoline　during　the　Worid　War　l　l　in　USA．　The　cracking　of　vaporized　oil　and

reactivation　of　the　spent　catalyst　in　the　regenerator　are　continuously　performed　in　the

process，　where　byproduced　ooke　is　bumt　to　maintain　the　range　of　reaction　temperatu　re

from　500　to　800　K．

　　　　Here，　practi’cal　features　of　CGFB　system　are　briefly　summarized　below：

1）　High　gas　ve［ocines　compared　with　those　in　conventionat　bubbling　fluidized　beds：

　　a）　Employment　of　various　solid　panicles，　including　fine　and　coherent　particles

　　b）　Plug　flow　of　fluidizing　gas，　in　other　words，　adjustable　residence　time　up　to　severel

　　　　seconds

2）　High　slip　velocity　between　gas　and　solid　particles：

　　a）　Promotion　of　heat　and　mass　transfer　between　them

　　b）　Enhancement　of　mobility　of　solid　particles

3）　Others：

　　a）　Temperature　uniformity　in　the　riser

　　b）　Easy　control　of　solid　residence　time　by　changing　downoomer　solid　inventory

　　c）　Staged　addit’ion　of　gaseous　reagents　at　various　locations

　　d）　High　solid　loading　operation

Hence，　CG　FB　system　has　been　extensively　adopted　in　various　industrial　processes　such

as　separation，　drying，　adsorption，　coating，　granulation，　calcination，　combustion　and　so　on．

The　current　status　of　industrial　apPlications　of廿ie　system　is　summarized　in　Table　1．1．
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卜

ln　devebpment　of廿）ese　processes，　however，　the　following　disadvantages　have　been

indicated二

）a

ラ
　
　
　

　

Too　highly　sensin’ve　change　in　fluidiration　states　in　the　riser　to　operating　variat）les

and　propenies　of　solid　and　gas

Necessity　of　high　risers　to　give　a　sufficient　residence　time　for　gas　and　solid　panicles

Need　of　efficient　separation　of　fine　partides　and　dust　from　gas　flow

Unknown　and　undesirable　dependency　of　stable　and　flexible　operation　on　system

configuration

　　　　Most　of　th　e　above　advantages　and　disadvantages　would　reflect　the　complicated

hydrodynamic　charaCteristics　of　gas　and　solid　particles　in廿1e　riser，　which　are　arisen　from

their　turbulent　interactions　at　gas　veloctties　much　higher　than　the　terminal　velocity　of　the

single　panicle．　However，　in　spite　of　increasing　in　number　of　industrial　applications，　existing

methods　of　design，　control　and　scale－up　are　purely　empirical　and　almost　covered　by

engineering　know－how　at　various　stages．　This　is　due　to　a　lack　of　fundamental

understanding　of　unstable　and　heterogeneous　nature　of　gas　and　solid　flow　in　CGFB

system，

1　．2．2　Fiuidizati　on　States　in　Riser　of　CGFB　System

　　　As　briefly　described　in　1．1，　solid　particles　are　always　fluidized　aggregatively　by　gas．

At　IQw　gas　velocities，　gas　bubbles　form　above　the　distributor　and　rise　through　the　bed

growing　by　coalescence，　which　eventually　result　in　bulk　solid　circulation　and　rapid　solid

mixing．　［n　the　riser　of　CG　FB　system，　where　the　gas　velocity　is　usually　much　higher　than

the　parijcle　terminal　velocity，　particie　clusters　appear　near　its　bottom　and　rise　at　fairiy　high

velocities．　The　clusters，　depending　on　the　superficial　gas　veiocity，　ug），　and　the　circulating

solid　mass　flux，　G，，　sometimes　disappear　along　the　height　forming　a　dilute　phase　and

consequently　giving　an　S－shaped　bed　density　distribution．　This　sort　of　flow　pattem　is

affected　in　a　highly　complex　manner　by　configuration　of　CGFB　system　as　well　as廿）e

above　operating　variables．　ln　the　last　two　decades，　numeroUs　investigations　have　been

Canied　out　to　characterize　the　flow　pattem　in　the　riser　from　macroscopic　and　m　icroscopic

observations　with　various　experimental　methods．　They　are　briefly　reviewed　in　the

following．

　　　Macroscopic　flow　pattern　i　n　CGFB　riser　is　to　a　large　extent　affected　by　ugp　and　Gs，

Which　in　tum　offers　a　wide　flexibility　in　handling　of　solid　particles　（Yerushalmi；　1986）．　So

　　　　　　　　　　　　　　　　　　　　　　　　　　　　－11一
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far，　several　investigations　（Yerushalmi　and　Cancurt；　1976，　lj　and　Kwauk；　1980，　Rhodes；

1989，　Takeuchi　et　al；1　986，　Horio　et　al；　1986）　have　attempted　to　classify　the　flow　regime　for

better　understanding　of　hydrodynamic　characteristics　in　CGFB　risers．　Yerushalmi　and

Cancurt　（1979）　investigated　variation　of　th　e　static　pressure　gradient，　d　P／dz，　with　G，　at　a

certain　height　in　the　riser　for　fixed　gas　velocities．　Typicai　variations　are　shown　in　Fig．1．4．

As　can　be　seen，　a　rather　smooth　increase　of　d　P／dz　with　Gs　is　seen　at　high　gas　velocines

while　an　abrupt　increase　appears　at　relatively　low　gas　velocde’es．　This　abrupt　increase

refiects　occurrence　of　transition　of　the　fluidization　state　in　the　riser　from　a　dilutephase

transport　state　to　a　dense－phase　transport　state　and　becomes　indistinguishable　at　upo

higher　than　a　critioal　velocity．　This　critical　velocity　was　defined　as　the　transport　velocity，　utr．

Furthermore，　the　state　of　fiuidization　at　upe　h　igher　than　utr　was　named　as　fast　fiuiditation，

where　the　gas一一solid　slip　velocity　varies　widely　wth　ucp　and　Gs．　U　and　Kwauk　（1980）

defi　ned　the　fast　fluidization　state　differentiy　from　the　above　as　a　state　that　exhib’rts　an　S－

shaped　axial　bed－dens’rty　disbibution　where　lower“dense　and　upperdilute　regions　are

formed　in　the　n’ser．　Similar　definitions　have　so　far　been　proposed　by　other　investigators

（Rhodes　and　Geldart；　1　986，　Takeuchi　et　al；　1　986）．

　　　　Based　on廿）e　above　definitions　of　the　fast　fluidization　state，　flow　regime　classification

maps　have　been　constructed　（Yerushalmi　and　Cancurt；1979，　li　and　Kwauk；　1980，

Takeuchi　et　al；　1986，　Horio　et　al；　1986，　Rhodes；　1989）．　They，　as　a　matter　of　course，　differ

one　another　due　to　their　inconsistent　definitions　of　the　fast　fluidization　state．　Figures　1．5

and　1．6　represent　the　ciassification　maps　proposed　by　Yerushalmi　and　Cancurt　（1　979）　and

Rhodes　（1989），　respectively．　ln　the　former　map，　utr　is　employed　to　give　a　boundary

between　the　turbulent　and　fast　fluidization　states　and　the　region　of　fast　fluidization　is

classified　in　a　range　of　gas　velocity　from　utr　to　a　certain　upo　at　which　the　maximum　slip

veloc’rty　is　obtained．　I　n　the　latter，　on　the　other　hand，　the　region　is　classifiecl　between　the

densephase　transport　and　dilutephase　transport　states　with　solid　reflux．

1．2．3　Effects　of　System　Configuration

　　　　As　mentioned　above，　the　flow　pattems　in　the　riser　of　CGFB　system　are　known　to

depend　on　’唐凾唐狽?ｍ　configuration．　Nonetheless，　lntle　attention　ha’刀@been　paid　to　the

dependency　and　there　have　been　a　limited　number　of　investigations　of　its　effects　on　the

axiEtl　bed－dens’rty　（ABD）　distribution　（Brereton　et　al；　1994，　Zheng　and　Zhang；　1994，　Hirama

et　al；　1992）．　Figure　1．7　shows　a　representative　example　for　effect　of　the　shape　of　riser

exit　on　the　ABD　distribution　（Brereton　et　al；　1994）．　lt　is　apparent　from　the　figure　that　the

disbibution　over　the　whole　riser　is　oonsiderably　affected　by　the　exit　shape．
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　　　　It　is　also　known　that　pressure　difference　between　the　riser　and　downcomer　bottoms

has　an　appreciable　influence　on　stabil’rty　of　solid　circulation　in　CGFB　system　（Takeuchi　et

al；　1986，　Bi　and　Zhu；　1　993）　when　a　pneumatic　valve　like　an　L－shaped　or　a　loop－seat　valve

is　employed　as　a　solid　feeding　device．　Hirama　et　al　（1992）　conducted　a　measurement　of

the　static　pressure　distribution　in　the　riser　at　various　ugp　and　Gs　（Fig．1．8）　l　n　order　to

evaluate　the　effects　of　solid　feeding　device　on　the　disbibution，　they　utilized　two　different

solid　feeders；　a　hopper－screw　feeder　system　which　controls　uge　and　Gs　independentiy　of

the　pressure　difference　and　a　conventional　ball　valve　feeder　（Takeuchi　et　al；　1986）．　They

found　that　ABD　distributions　vvith　the　former　solid　feeding　device　were　purely　determined　by

uge　and　Gs　and　proposed　a　macroscopic　flow　regime　classification　map，　which　was

constn」cted　on　ug）く∋s　diagram　as　shown　in　Fig．1。9．　However，　as　shown　in　Fig．1．10，　Li　et

al．　（li　et　al；　1988）　observed　that　th　e　ABD　distribution　was　found　to　depend　on　the

downcomer　solid　inventory　when　a　pneumatic　valve　is　employed　as　a　solid　feeder　and　the

pressure　difference　is　affected　by　the　downcomer　solid　inventory．　．
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reMts1’

1　．2．4　Slip　VeEocity　in　CGFB　Riser

　　　One　of廿le　charaCteristic　features　of廿）e　pa面cle　flow　in　a　CGFB　riser　is　a　high　slip

velocity　between　fluidizing　gas　and　particles．　Here，　the　apparent　gas－solid　slip　velocity，

usip，　is　expressed　as：

usip　＝　uge　／E一　Gs　／　ppt（1　一　E） （1－1）

where　pp　and　E　respecxively　denote　the　particle　dens’rty　and　the　mean　bed　voidage．　Lewis

et　al（1949）first　reported　usip　observed　at　various　opera加g　oond面ons　in　a　CGFB　riser　and

showed　that　uso　is　considerably　higher　than　the　terminal　velocity　of　single　particle，　lh，　when

fine　particles　are　employed　as　a　bed　material．　Yerushalmi　et　al　（1979）　examined　usip

under　a　wider　range　of　uge　and　Gs，　and　found　that　usip　possibly　attains　more　than　thirty

times　of　ut　（Fig．1．11）．　They　suggested　that　such　a　high　slip　veloc’rty　can　be　expla’ned　by

assuming　formation　of　pafu’cle　clusters　having　terminal　velocities　higher　than　ut　and・

developed　a　model　to　predict　the　cluster　size．　・
　　　On　the　other　hand，　an　explanation　different　from　the　above　is　possible　based　of　the

radial　distribution　of　solid　mass　flux　in　a　riser　（Geldart　and　Rhodes；　1986b）．　The　radiat

disbibutions　frequently　exhibit　the　so－called　”core一一annulus”　flow　structure　oonsisting　of　an

upward　dilute　flow　in　the　center　and　a　downward　dense　fiow　near　the　vedical　wat1　of　the

riser　（Fig．1．12，　Rhodes；　1990）．　Furthermore，　Mineo　（1989），　measuring　the　interstitiat　gas

velocity　by　a　tracer　technique，　found　that　the　observed　gas－solid　slip　velocity　has　a　distinct

radial　distribou’on，　as　shown　in　Fig．1．13，　and　is　appreciably　lower　than　the　apparent　slip

veloc’р凵@calcuiated　by　Eq．1－1．　ln　ackiition，　usip　becomes　higher廿lan　ut　at　any　lateral

position，　except　near　the　boundary　between　the　core　and　annulus　regions．　Thus，　there

has　been　no　general　agreement　for　explanation　of　the　high　slip　velocities　observed　in

CGFB　risers．

1・3　Circulating　Liquid－Fiuidized　Bed（CLFB）System

　　　In　comparison　wth　CG　FB　systems，　1ntle　has　been　clarified　of　flow　characteristics　of

solid　particles　fluidized　with　liquid　at　velocities　higher　than　the　terminal　veloc’rty　of　single

Particle　and　most　of　the　existing　findings　have　been　obtained　in　batch　liquid－fluidizeci　bed

syste　m．

　　　As　mentioned　in　1．12，　solid　particles　are　particulately　fluidized　by　iiquids　uniess　the

dens’rty　difference　between　the　solid　and　liquid　is　not　so　big．　Partlcles　are　homogeneously

SUSpended　in　liquid　stream　and　the　particle　bed　expands　wlth　an　increase　of　the　liquid

　　　　　　　　　　　　　　　　　　　　　　　　　　　　－21一
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velocity．　lf　the　loganthms　of　the　bed　voidage，　q　are　plotted　against　those　of　the　supenicial

tiquid　velocity，　ub，　a　linear　relationship　is　obtainable　between　them，　which　can　be

extrapo［ated　to　E　＝　1　at　ub　approximately　equal　to　ut　and　generaliy　described　by　the

following　equation：

ub／　ut　＝　En （1－2）

Lewis　and　Boweman（1952）first　repo段ed　expehmental　data　reiating　to　the　above

relationship　and　later　Richardson　and　Zaki（1954）determined　the　empihcal　oonstant，　n，　in

the　above　equation　for　various　of　particles．

　　　When　particle　bed　is　fluidized　at　ub　higher　than　u｛，　a　lほhe　pa淀icles　are　entrained　by

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　ノ
liquid　from　the　top　of　the　column．　If　they　are　r㏄ycled　to　the　bottom　of　the　Golumn　as

usualiy　seen　in　CGFB　system，　a　certain　voidage　less　than　un詮y　is　attainable　h　the　column．

According　to　a　few　existing　findings，　high－velocity　fluidization　charactehstics　in　the　riser　of

circulating　Iiquid・fluidized　bed（CLFB）system　seem　to　be　much　simpler　and　more　stable

than　those　in　CGFB　hser　which　usually　exhib仕aggregahve　phenomena　such　as　the　bubble

and　cluster　fomation，　the∞re－annuius　flow　strudure，　the　S－shaped　ABD　dis恒bu面on　and

the　system　hstabil忙y　ahsen　from　the　system　configuration．

1　．4　Scope　of　Present　Study

　　　The　present　study　aims　to　elucidate　flow　pattems　of　solid　particles　fluidized　at

velocities　higher　than　the　terminal　velocity，　i．e．，　those　in　the　riser　of　CGFB　and　CLFB

systems．　Attention　has　consistentiy　been　focused　to　effects　of　operating　variables　such

as　the　riser　height　and　the　ini廿al　solid　inventory　in廿〕e　downoomer　on　ABD　disbibutions，

which　have　been　obtained　from　measurements　of　the　static　pressures　along　the　height　of

the　riser　and　downcomer　in　a　CGFB　system　having　5．5　m　and　10　m　high　risers　with　an

inner　diameter　of　O．1　m．　I　n　order　to　simplify　the　complicated　and　unstable　flow

characteristics　in　the　CG　FB　system，　a　circulating　liquid－fluidized　bed　（CLFB）　system　with　a

4．0　m　high　and　O．049　m　i．d．　riser　were　employed　and　the　systematic　measurements　were

ca市ed　out　for廿ie　eD〈ial　static　pressure　distribution廿1rough　the　n’ser　as　well　as　fluCtuations　in

the　local　bed　voidage　and　conductive　heat　flux，　all　of　which　strongly　reflect　parb’de　flow

pattems　in　the　riser．
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Chapter　2

EFFECTS　OF　RISER　HEIGHT　AND　DOWNCOMER　SOLID　INVENTORY

　　　　　　　　　　ON　ABD　D量STRIBUTION　IN　CGFB　RISERS

Zl　lntroduodon

　　　As　introduced　in　the　preVious　chapter，　axial　bed　density（ABD）distribし甫ons　in　a

CG　FB　riser　have　been　measured　by　many　investi’№≠狽盾窒刀@and　the’　flow　regime　classMcation

maps　h　ave　been　proposed　on　th　e　basis　of　th　e　variation　of　ABD　distrl’btm’ons　with　ug3　and　Gs．

This　is　because　ABD　distributi’ons　is　obtained　as　a　resutt　of　macroscopic　flow　characteristics

which　directiy　reflect　fundamentEti　ch　aracteristi’cs　of　fl　u　idizati　on　state　of　solid　panicles　at　high　．

gas　velocin’es．　ABD　distn’butions　have　been　found　to　depend　not　only　on　operating

varial”es　but　also　on　oonfiguration　of　the　solid　circulation　system．　However，　in　existing

investigations，　l　nve　attention　h　as　been　focused　to　the　latter　effect．

　　　Most　of　ABD　disbibut）’ons　in　l　iteratu　re　were　obta’ned　in　CGFB　risers　where　the

downcomer　bottom　is　simply　oonnected　to　the　riser　bottom　with　L一一shaped　or　loop　seat　velve

which　usuaiiy　has　a　pressure　drop　comparable　to　or　smatler　than　that　through　the　riser　and

the　downoomer．　Th　us，　with　such　a　solid　feeding　device，　pressure　interact）’on　would

inevitably　be　induced　between　the　riser　and　downcomer，　resuhing　in　ABD　distributi’ons

which　more　or　less　depend　on　experimental　system　oonfiguration．　For　instance，　some

investigators　reported　that　ABD　distributions　vary　appreciably　with　the　weight　of　so　l　id　i　nin’at　ly

loaded　in　the　downcomer　（downcomer　solid　inventory）　even　at　a　given　set　of　the　gas

veloclty，　uge，　and　the　circulating　solid　mass　flux，　Gs，　（Weinstein　et　al；　1983，　li　et　al；　1988，

Mori　et　at；　1989）．　However，　it　has　never　been　made　dear　the　oorrespondence　between

the　distribution　and　the　press　u　re　i　nteractjon．

　　　Hirama　et　al　（1992）　made　a　systematic　measurement　of　A’BD　distributi’on　in　an

experimented　CG　FB　riser　h　avi　ng　a　hopper‘screw　feeder　with　which　ugo　and　Gs　were

lndependenゼy　controlled　supPressing　Pressure　intera（：ガon　between　廿1e　riser　and　the

down　comer　bottoms．　The　resu　tts　evidenced　from　the　measurement　are　as　follows：

1）　S－shaped　ABD　distributi’ons　as　a　resuit　of　formation　of　the　lowerdense　and　upper一

　　　　　　　　　　　　　　　　　　　　　　　一28一
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　　　　　dilute　regions　are　obtEti’ned　at　relatively　low　ugo　and　Gs　but　are　never　observed　when

　　　　　uge　and　Gs　are　higher廿1an　the　scHンalled　ilranspOrt　velocity’，　and　corresponding

　　　　　circulating　solid　mass　flux．

　　2）　ABD　distil’btm’on　is　essentially　independent　of　the　downoomer　solid　inventory　when

・　　　no　pressure　interactjon　exists　betWeen廿ie　riser　and廿1e　downcomer．

Based　on　these　findings　together　with　existing　models　for　solid　fiow　in　the　freeboard　of

bubbHng価dized　beds，　a　conceptUat伺ow　regime　diagram　was　construded　simply　as　a

funcljon　of　uge　and　Gs．　However，　their　ABD　distribut’ons　were　measured　at　a　fixed

downcomer　solid　inventory　and　its　effects　on　their　diagram　were　not　examined．　Th　us，　in

this　chapter，　the　effect　of　the　pressure　i　nteraction　on　ABD　distribution　is　examined　on　the

basis　of　systematic　observations　for　a　range　of　the　clowncomer　sdlid　inventory．

22　Experi　menta　l

　　　fi　g　u　re　2．1　i　llustrates　a　schematic　diagram　of　CG　FB　system　employed　here．　Two．

dnterentheight　risers，　CG　FB55　and　一1　OO，　were　employed　which　have　respective　heights

of　5．5　m　and　1　O．O　m．　Beth　risers　were　made　of　a　transparent　PMMAiesin　column　having

an　inner　diameter　of　O．10　m．　Pressure　taps　were　attached　vedically　on　the　riser　wedl　at

intervats　from　O．1　5　to　O．55　m　for　measurements　of　the　static　pressure　distn’bution　in　the　riser．

As　a　moving－bed　downcomer　a　O．15　m－dia．　oolumn　was　used　for　CGFB－55　system　while

for　CGFB－100　system　a　020　mdia．　column　was　used．　Both　of　the　oolumns　have

pressure　taps　attach　ed　on　their　wel　1．　Each　down　oomer　bottom　was　simply　connected　wtth

the　riser　bottom　by　a　screw　feeder．　The　feeder　oontrols　the　a’rculating　mass　flux　of　solid

partides　by　adj　usting　revoltm’on　speed　of　the　screw．　The　fl　uctu　ation　in　the　solid　feeding

rate　per　revolu百on　was　less廿1an　1％of廿1e　average　rates．　Differential　pressu　re　between

two　different　venical　pos’rd’ons　in　the　riser　were　measured　by　water　manometers．　The

measurements　of　th　e　press　u　re　distri　but’on　were　conducted　at　steady　state　wh　ich　was

confirmed　by　measuring　the　totat　pressure　drop　through　the　riser　by　an　electrk）　pressure

trarisducer．

　　　All　experimentS　were　cam．ed　oUt　at　ambient　cond市ons　using　air　as　a　fluid’rzing掘d．

The　bed　material　used　was　FCC　panicles　with　a　median　of　54　pm　in　the　padide　size

distn’but）’on　and　with　a　mercuryporosimebic　density　of　approximately　810　kg・mB．　The

SUPediciat　gas　velocity　and　the　circulating　solid　flux　in　the　present　experiments　ranged　from
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P　；　pressure　tap

1．　riser　（5．5　＆　10m　height，　O．1m　i．d．）

2．　primary　cyclone

3．　secondary　cyclone

4．　downcomer
5．　screw　feeder

6．　bag　filter

Fig・2・1　Schematic　diagram　of　experimental　circulating　gas－fluidized　bed　system
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1　．3　to　3．0　m・s－i　and　4　to　60　kg・m’2・s’‘，　respectively．

2．3　Resutts　and　Discussion

23．1　Pressure　Disセibu市on　in　Circulating　System

　　　Figure　2・2　shows　typical　pressure　disbibutions　observed　in　CGFB－55　system．　The

disthbu廿ons　were　obtained　at　a行xed　u⊆p　of　1．3　m・ざ1　and　Gs　of　20　kg・m’2・s’1　for　different

sta廿。　bed　heights　in’hiatly　loaded　in　the　downoomer，　Hsd，　of　1．8，2．3　and　2．7　m．　W㎞

increasing　Hsd，　the　statb　pressure　at　the　bottom　of　the　downoomer，　Pα），　simply　increases

WithoUt　affecting　the　pressure　at　the　riser　bOttom，　Pゆ，　and　the　pressure　disbibLrtjons　in　the

riser．　Though　not　shown　here，伺u（；ti」ation　was　observed　in廿le　soiid　feeding　rate　when　P由

became　h　igher　than　Prひ　ln　this　case，廿1e　stable　solld　feeding　becam　e　impOssible　due　to

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　ノ
the　significant　pressure　interacb’on　between　riser　and　downcomer　bOttoms．　Hence，　ABD

disbibut）’ons　shown　hereafter　were　measured　at　condin’ons　where　Pro＞P（b。

2．32　Change　i　n　Loca　l　Bed　Density　’
　　　Figure　23　oompares　ABD　distributions　obtained　at　van’ous　u｛p　and　Gs　in　CG　FB55

riser．　At　higher　u．　（Fig2．3（c）　＆　（d）），　the　particle　bed　atvvays　exhibits　homogeneous　ABD

dis廿ibし而on　in　the　riser，　except　for　the　inlet　and　oし雨et　regions．　As　can　be　seen　in廿1e　figure，

bed　voidage　monotonically　decreases　with　increasing　in　Gs　within　the　present　range　of　Gs．

This　deaily　indicates　that　the　fluidization　state　in　such　range　of　ub　changes　from　the　dilute－

phase　transport　to廿1e　dens（耳）hase　transport　oontinuously　with　Gs。　At　lower　ug◎

（Fig・2．3（a）＆（b）），　on　the　other　hand，廿1e　voidage　abruptiy　decreases　at　a　pan廿cular　Gs（e．9．

Gs　＝　20　kg・m－2・s－i　for　uge　＝　1　．5　m・s’i）　at　wh　ich　so－called　S・一shaped　ABD　distribub’on　appears

in　the　riser．　　This　refie（尤3　the　sudden　trans詮jon　of廿1e　fluidization　state　occurs　at廿1e

parb’cular　Gs．　Here，　it　should　be　noted，　as　drawn　by　broken　lines　in　Fig．2．3（a），　that　ABD

disbibution　trans’de’vely　changed　with　time　even　at　fixed　upe　and　Gs　when　Gs　takes　a

Partioular　value　and　the　u　nsteady　phenomena　have　been　discussed　later．　Figure　24

Shows　ABD　distrt’bution　in　CG　FB－100　which　were　obtEained　at　the　same　operating

condin’ons　for　CGFB55　（Fig2．3）．　lt　becomes　clear　from　the　figure　that　the　ABD

distnbutjon　becomes　homogenbous　at　higher　upe　while　the　fiuidization　state　suddenly’

Changes　with　increasing　in　G，　at　Iower　uge，　as　observed　in　the　lower　riser．　The　tansdion　in

the　m　aoroscopic　fl　u　idization　states　is　fu　rth　er　exam　ined　from　the　variation　of　the　loced

PreSsure　gradient　at　a　certain　height　of　the　riser．　l　n　Fig．　25，　the　local　bed　density　in　terms

Of　Static　pressure　gradient　一dP／dz，　at　a　certain　height，　z，　of　the　risers，　i．e．，　z＝　2．4　m　for
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cG　FB－55　and　z　＝　72　m　for　CGFBI　oo，　respectively，　was　plotted　against　G，　at　various　gas

velocftjes．　For　all　cases　the　bed　density　increases　wtth　Gs．　No　d’nterences　are　seen

between　the　bed　denshies　in　CGFB－55　and　一1　oo　risers　partiou　lariy　when　G，　is　small　and

even　the　differences　seen　at　higher　Gs　are　not　essential　with　in　the　present　experimentat

accuracy．　As　Gs　increases，　an　abrupt　change　in　the　bed　density　o（x　u　rs　at　upe　lower　than

1．7s　m・s’i　suggesting　the　development　of　the　dense　region　in　the　riser　and　transin’on　of

fluidization　state　from　the　dilutephase　transport　state　to　the　densephase　transport　state．

At　lower　gas　velocin’es，　the　transib’on　o（x）u　rs　at　a　lower　range　of　Gs．　The　transkion

becomes　ambiguous　as　ugp　increases　and　seems　to　disappear　when　uge　becomes　arou　nd

2．0　m．s－i．

　　　From廿1e　observed　curves　of－dP／dz　vs．　Gs，廿）e　gas　veloc仕y　corresponding　to廿le　so－

c創led施nsport　veわcdy～t．lt，・，　de伽ed　by　Yerushalmi　and　Cancurt（Yerushalmi　and　Cancu　rt；

1979），　was　evaluated　approximately　as　2．0　m・s一‘．　Th　is　implies　that　the　S－shaped　ABD

distriヒ＞utions　are　expected　to　apPear　at　gas　veloc面es　less　than　this　par廿cular　gas　velocity　for

廿le　present　soiid　pa而cles．　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　・

23．3　Change　in　Height　of　Dense　Region

　　　Figure　2．6　shows　changes　with　time　of　the　total　pressure　drop　th　rough　the　riser，

APtr　ta，　measured　in　CGFB55　riser　at　upe　of　1．3　m・s一’　as　a　fu　ncb’on　of　Gs　and　Hsd．　lt　shou［d

be　noted　that　u　n　steady　decrease　of　A　Pma　is　observed　at　Gs　＝　15　kg・m’2・s’i．　For　Hsd’s　of

1．8　and　2．1　m，　APma　gradually　decreases　wtth　time　under　oond｛nions　where　the　pressure　at

the　n’ser　bottom，　Pro，　is　higher　than　that　at　the　downoomer　bottom，　Pde，　and　an　approxl　mate

steady‘state　is　attained　afte　r　P．　beco　m　es　n　eariy　equal　to　Pbo．　At　Gs　of　10　and　20　kg・m’2・s’

1
，　on　th　e　oth　er　hand，　A　Pna　ata　ns　a　steady　state　fairiy　quickiy　after　ugo　and　Gs　are　fixed．

This　quick　response　was　obviously　independent　of　Hsd．

　　　The　unsteady　change　in　APma　oorresponded　not　only　to　the　drastic　increase　in　the

Pressure　gradient　with　Gs　described　i　n　Fig2．5　but　also　to　signlficant　changes　in　height　of廿1e

dense　region　height　which　was　simultaneously　observed　at　the　condin’ons　indicated　as　a，　b，

C，　d　and　e　in　Fig2．6．　The　results　are　shown　in　Fig．2．7　as　ABD　distributions．　At　Gs　＝　15

kg・m－2－s’i，’
≠刀@drawn　by　broken　lines，　the　S－shaped　ABD　distn’butjons’ ≠窒?　seen　at　condnions

of　a，　b，　c，　d　and　e．　lt　is　also　seen　that廿ie　height　of　the　lower　dense　region　decreases　with

APtota　or　with　Hsd　as　described　i　n　Fig．2．6　even　at　fixed　uge　and　Gs．　From　th　e　resutts　shown

ln　Figs・from　2．5　to　2．7，詮is　eVident　that廿1e　eflect　of　Hsd　is　observed　only　at　a　pa而cular

cond市on　of　Gs　and　ug，　Where　the　S－shaped　ABD　distributi’on　is　obtained　in廿）e　riser．

　　　The　above－described　experimental　results　oould　be　expla’ned　by　an　analogy　to　solid

　　　　　　　　　　　　　　　　　　　　　　　　－36一
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entra’nment　in　the　freeboard　of　the　bubbling　fluidtzed　beds　（Kunii　and　Levensplel；　1990，

Rhodes；　1989，　Wen　and　Chen；　1972，　Zenz　and　Othmer，　1960）．　Figure　2．8　illustrates

typiceil　disbibutions　of　upward　so［id　fi　u　x　along　the　height　from　the　bottom　of　a　riser　vvith　a

height，　Hr．　The　solkl　mass　fl　ux　decreases　with　height　exponentiatly　from　a　fl　ux　at　the

dense－region　surface，　Fo，　to　a　partiou　lar　fiux，　F．．　via　a　transta’on　region，　Htra，s．　［f　the　dense－

region　height　is　here　defined　as　the　height　at　which　a　decrease　in　the　solid　flux　commences，

the　height　of　the　denseイegion　oould　be　changeable　from　zero　to　Hdpaax　which　is　equEal　to　Hr

一ト1幡．

23．4　Effect　of　Riser　Height　on　Dense一　Region　Height

　　　ABD　disbibutions　for　both　CG　FB　risers　at　ug3　＝　2．0　m・s－i　for　various　Gs　ranged　from

lo　to　60　kg・m2・s’i　are　oompared　in　Fi　g．2．9．　Here，　the　distri　bution　cu　rves　are　drawn　by

leveling　the　exit　for　both　risers．　As　can　be　seen，　the　distn’butions　of　the　two　risers　seem　to

be　similar．　l　n　addin’on，　when　an　S－shaped　ABD　distributions　appear　as　observed　at　Gs　＝

40　and　60　kg“m2s’i　an　obvious　expansion　of　the　dense－region　is　seen　for　the　taller　riser．　・

24　Conclusion

　　　Measurements　of　ABD　disbibution　were　oonducted　in　risers　of　durerent　heights　with

the　same　diameter　of　CG　FB　system．　The　effects　of　the　riser　height　and　downcomer　solid

inventory　on　the　disbibutions　were　examined．　Within　the　present　experimental　oonddion，

the　followings　were　found；

1）　When　the　static　pressure　at　the　riser　bottom　is　higher　than　or　equal　to　that　at　the

　downoomer　bottom，　the　ABD　distributions　in　the　riser　at　a　fixed　gas　velocity　and　solid　fiux

・　are　generally　independent　of　the　tota1　solid　inventory　in　the　downoomer．

2）　At　a　pardcular　G，，　the　dense－reglon　height　signMcantiy　depends　on　the　downoomer

　solid　inventory．　The　resutt　can　be　explained　by　an　anatogy　to　existing　models　of　the

　solid　elしrb『ation　in廿）e　freebOard　of　the　bubbling　fluidized　bed．

3）　Denseregion　height　increases　with廿）e　riser　height　even　under　oondhions　where　a　ll廿le

　other　operating　variables　are　the　same．　H　ence，　an　apparent　overiap　is　obtained　for　the

　ABD　dist］1’buti’on　curves　for　hsers　of　the　d群erent　heights，　when廿1e　distribし市ons　are

　drawn　by　leveling廿1e　riser　exit．
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Chapter　3

　　　　　UNSTEADY　STATE　BEHAVIOR　OF

A　CIRCULATING　GAS－FLUIDIZED　BED　SYSTEM

3．1　lntroductr’on

　　　A　circu　lating　gas－fluldtzed　bed　（CGFB）　system　basically　consists　of　a　riser，　a

downoomer，　cyclones　and　a　solid　feeding　device．　Panides　elutn’ated　from　the　riser　top　are

collected　by　the　cyclones　and　fed　to　the　downoomer．　Then，　the　panicles　descend　through

the　downcomer　and　continuously　fed　th　rough　the　solid　feeding　device　to　the　riser　bottom　at

a　steady　mass　flux．　The　stable　solid　circu　lation　is　known　to　depend　on　the　pressure　drop　．

through　the　feeding　deVice，△Prd．　Most　of廿1e　axial　bed　dens辻y（ABD）distributions　in　the

riser　reported　in　1iteratu　re　were　observed　in　experimentaJ　CGFB　systems　where　the

downcomer　bottom　was　simply　oonnected　to　the　riser　bottom　by　an　L，shaped　or　a　loop－

seat　vaEte．　l　n　the　case　of　CG　FB　system　using　such　a　vatve，　the　observed　steady－state

ABD　distributions　were　found　to　be　dependent　more　or　less　on　the　inin’al　downcomer　solid

inventory　even　at　fixed　gas　velocity，　uge，　and　circu　lating　solld　fi　ux，　Gs　（VVeinstein　et　al，　1983，

U　et　al，　1988，　Mori　et　al，　1　989）．　This　is　attn’buted　to　smatl　APrd　of　the　pn　eumatic　vabee．　l　n

fact，　solid　circu　lation　is　sensin’ve　to　A　Prd，　taking　a　longer　period　for　attainment　of　a　steady

state　due　to　a　pressure　interaction　between　the　riser　and　downcomer　bottoms．　lt　has　atso

been　observed　that　an　apPredable　fluc恵」a廿on　in廿1e　ABD　distlibし甫on　is　brought　about

during　the　transin’onal　period　for　attatnment　of　the　pressure　belance　in　the　whole　system．

However，　in　the　existing　investigations　the　circulating　solid　mass　fiux　has　never　continuously

been　measured　and　details　have　not　been　clartfied　yet　of廿ie　unSteady・state　charaCteristics

Of　the　CG　FB　system，　such　as　G，，　APrd　and　the　solid　holdup　in　the　riser　and　the　downcomer．

　　　ln　the　preceding　chapter，　a　screw　feeder　having　APrd　much　larger　than　pneumatic

Vatves　was　employed　to　minimize　effect　of　the　variation　of　the　downoomer　bottom　pressure

On　ABD　distn’butions　（Hirama　et　at，　1996）．　From　measurements　of　ABD　distributbns　for

d‘nterent　heights　of　the　static　bed　of　panicles　inib’ally　loaded　in　the　downcomer，　Hsd，　it　was

fOUnd　that　the　ABD　distn’bution　depends　on　Hsd　only　at　odnd’de’ons　where　the　dense　region

aPPears　in　lower　part　of　the　riser．　However，　again，　deta’ls　are　stj　l　l　in　questi’on　conceming

　　　　　　　　　　　　　　　　　　　　　　　　　　－43一
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the　solid　flow　characteristics　in　the　circulating　system，　in　partiouley’，　the　change　of　the　static

ptessures　with　time　at　the　riser　and　downcomer　bottoms　wh　ich　g’Nes　rise　to　unsteady．state

ABD　distributions　in　the　riser．　Hence，　in　this　chapter，　a　further　experimental　work　has　been

canied　out　to　eval　uate　the　relation　of　th　e　variation　of　APrd　to　the　ABD　distribution　from

continuous　measurements　of　the　static　pressures　as　well　as　Gs　for　various　Hsd．　l　n　addin’on，

the　effect　of　static　pressure　at　the　downoomer　bottom　on　steady．state　ABD　distribution　was

examined　var＞，4’ng　intentionally　the　downcomer　bottom　pressure　by　aeration．

3．2　Experi　menta　l

　　　Figure　3．1　illustrates　a　schematic　diagram　of　the　experimental　CGFB　system．

Detatls　of　the　elements　in　the　system　and　experimentat　procedure　were　specified　in　the

preceding　chapter．　The　bottom　of　a　10．O　m　high　riser　with　an　inner　diameter　of　O．1　m　was

oonnected　to　that　of　a　02　m　i．d．　and　8．2　m　high　downoomer　by　a　screw　feeder　which

oontrolled　the　circulating　mass　flux　of　solid　panicles，　Gs，　by　adjusting　a　revolution　speed　of

the　screw　without　aeration．　Changes　with　time　of　the　axiat　distn’butions　of　static　pressure　in　s

the　riser　and　the　downoomer　were　simultaneously　measured　with　electric　transducers

which　were　attached　on　the　riser　and　cic）wncomer　walls　at　respective　intervals　from　O．45　to

O．60　m　and　from　O．40　to　1．00　m，　respectively．　Signats　from　the　transducers　were

processed　by　a　personal　oomputer．　At　a　oondinon，　about　20　seconcls　were　required　to

obta’n　data　for　determination　of　ABD　distil’but）’on　in　the　riser．　As　shown　in　Fig．3．2，　an

impact－type　solid　flow　device　（linoya　et　al，　1966）　was　presentiy　employed　to　monitor　the

flow　rate　of　solid　panicles　elmiated　from　the　top　of　the　riser．　The　device　was　eouipped

immediately　below　the　primary　cydone．　The　solid　mass　flow　rate　was　oontinuously

measured　by　a　load－cell　wh　ich　detects　the　venical　component　of　impact　force　exerted　by

廿ie　solid　par廿cles　on　a　declined　plate．

　　　The　bed　material　used　in　the　present　experiment　was　FCC　panicles　having　a　median

Of　54　pm　in　the　size　disbibution，　a　mercu　ryporosimebic　density　of　about　810　kg・mB　and　the

minimum　fluidization　velocity　of　O．0011　m・S’1。　The　par哲des　were価dized　by　air　at　r∞m

temperature　and　a　fixed　supenictal　gas　velocity　of　1　．3　m・s一‘　in　the　present　experiment．

3・3　Results　and　Discussion

3t3・　1　Tra　ns　’ta’onal　Change　of　Static　Pressu　res　at　Riser　and　Downcomer　Bottoms

　　　When　the　static　bed　height　of　par七cies醐aHy　loaded　in　the　downcomer，　Hsd，　was　less

than　2．7　m，　it　was　impossible　to　operate　the　present　CFB　system　steadily　due　to

OCCu　rrence　of　a　considerable　gas　cross　flow　between　the　riser　and　the　downcomer　through　’

　　　　　　　　　　　　　　　　　　　　　　　　　　一44一
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P　；　pressure　tap

1．　riser　（1　Om　height，　O．1m　i．d．）

2．　primary　cyclone

3．　secondary　cyclone

4．・impact－type　solid　flow　meter

5．　downcomer
6．　screw　feeder

7．　bag　filter

Fig・3・1　Schematic　diagram　of　experimental　circulating　gas－fluidized　bed　system
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the　screw　feeder．　Variation　i　n　the　static　pressures，　Pro　and　Pde，　at　the　riser　an　d

downcomer　bottoms，　respectively，　with　time，　t　elapsed　after　th　e　o　nset　of　the　solid

cirou　lation　（t　＝　O）　were　measured　as　a　fu　nction　of　the　cirou　Iating　solid　flux，　G，，　and　Hsd．

Figures　3．3（a）　and　（b）　compare　the　effect　of　Hsd　on　the　van’ations　of　Pro，　Pdo　and　their

dnterence，　APrd　（＝　Pro　一　Pcb），　at　Gs　＝　10　and　20　kg・m’2・s’i，　respectively．　From　visuat

cA　）servation　through　the　transparent　riser　watl，　the　panicles　in　the　riser　at　the　former　Gs　are

セansported　in　a　socadled“diluteT）hase　transporヒ’state　while　at　the　latter　Gs廿1ey　are

transported廿1rough　a　dense　region　over　the　riser，　as　indicated　in　the　preceding　chapter．　lt

iS　seen　in　the　figure　thaちat　both　Gs，　Po　tends　to　increase　very　quickly　after廿1e　onset　of廿1e

solid　ciroulation　and　to　reach　a　steady　value　within　a　short　time．　l　n　addin’on，　this　quick

van．％Non　of　Pro　w辻h　time　seems　to　be　hdependent　of　Hsd．　On　the　o廿1er　hand，廿1e

dependency　of　th　e　variation　of　Pde　on　Hsd　durers　wtth　G，．　At　G，　＝　10　kg・m2・s’i　（Fig．3．3（a）），

no　appreciable　variation　is　seen　i　n　Pcb　at　both　Hsd　and，　as　a　result　of　th　e　variations　of　Pro

and　Pcb，　a　steady　vaiue　of△Prd　is　attained　after廿1e　same　trans甫onai　periOd　as　th　at　for　Prc，

At　G，　＝　20　kg・m’2・s’i　（Fig．3．3（b）），　on　the　other　hand，　Pdo　apparentiy　varies　with　t　and　the’

variation　depends　on　Hsd．　At　Hsd　＝　2．7　m，　Pcb　increases　almost　lineardy　with　t　at　a　rate

slower　than　Pro　and　levels　off　at　t　＝　1．7　ks　whereas　at　Hsd　＝　32　m，　Pcb　increases　as　quickiy

as　Pro　and　after　attaining　a　ma＞q’mum　at　t　＝　1　ks　it　decreases　slowiy　and　falls　off　at　t　＝　6．8　ks．

At　these　condin’ons　Gs　quickiy　attaj’ns　a　prescribed　vat　ue．　Figure　3．4　represents　an

example　of　signal　from　the　impact－type　solid　flow　meter　recorded　for　G，　＝　20　kg・m’2・s’i　and

Hsd　＝　32　m．　lt　is　seen　that　Gs　increases　to　the　prescribed　vatue　within　1　．5　ks　despite　of　the

increase　of　APrd　till　6．8　ks　（Fig．3．3（b））．

　　　The　above－described　dependency　on　Hsd　of　the　dynamic　variation　of　Pde　at　Gs　＝　20

kg・m’2・s－1　was　exam剛by　ins麟．ng　unSteady・state　ve市（副di醐b而・ns。f愉e　s臨

pressure　in　the　downcomer，　Pd．　The　resutts　obtained　at　Hsd　＝2．7　and　32　m　are　shown　in

Figs．3．5　（a）　and　（b），　respect）’vely．　When　H，d　＝　2．7　m　（Fig．3．5（a））　the　distil’btm’on　of　Pd　at　a

Very　eaily　stage，　i．e．，　t　＝　O．Ol　ks，　is　neariy　uniform　with　a　slight　maximum　at　a　height　from

the　bottom，　Hd，　of　1．8　m．　As　ti　m　e　is　elapsed，　pd　i　n廿1e　Iower　part　starts　to　i　ncrease　with　t

due　to　a　small　amount　of　gas　flow　from　the　riser　to　the　downoomer，　which　was　visualy

observed　through　the　trtinsparent　waH　of　the　tube　between廿1e　screw　feeder　a：hd　the

CIOwncomer　bottom．　A　steady　state　distn’bution　is　attained　at　t　＝　1．6　ks　（Fig．3．3（b）），　which

svery　dose　to　that　at　incipient　fiuidiratjon　of廿）e　particles　as　is　shown　by廿1e　curve　at　t＝

3iO4　ks　in　the　figure．　When　Hsd　＝　3．2　m　（Fig．3．5（b））　such　a　distribution　is　seen　at　t＝　i．32

kS　but　Pd　in　th　e　lower　part　of　the　bed　（Hd　〈1．8　m）　then　decreases　1eaving　the　upper　part

dlstn’bし甫on　essentjaily　unchanged　and　the　steady　sセlte　distrbu廿on　is　obねlned　at　t＝6．8　ks，

　　　　　　　　　　　　　　　　　　　　　　　　　　一48一
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suggesting　that　the　bed　in　the　downcomer　oonsists　of　the　u　pper価dized　bed　and　the　lower

moVing　bed・The　formation　of　the価dized　bed（）an　be　a掘bu｛ed　to　an　upwad刊ow　of　gas

which　is　accompan　ied　with　partides　rEri　n　i　ng　down　from　the　cydones　onto　the　bed　surtace

（ti　et　al；　1994）．　Th　us，　it　is　clear　that　th　e　changes　of　Pro　and　Pde　with　time　in　the　circulati　ng

system　re刊ect　not　only　the　rapid　distn’buti’on　of　the　panicles　i而atly　loaded　in　the　downcomer

to　the　riser　and　the　downcomer　but　atso　the　slow　developm　ent　of　the　static　pressure

distn’butjon　in廿1こ口downoomer．　in　add甫on，　no　sign而cant　effect　of　the　variation　of　P（b　on

that　of　Pre　is　seen　from　the　figures．　Changes　h廿1e　sta哲。　pressures　w仕h価me　considerably

d群erent　from廿10se　deschbed　above　were　observed　at　G，＝15　kg・m皇S’1　for　all　Hsd，s．

Figure　3．6　shows　a　typicad　changes　of　Pro，　Pcb　and　APrd　observed　at　Hsd　＝　32　m．　A　sharp

inctease　is　seen　in　both　Pro　and　Pbo　as　soon　as　the　commence　of　solid　circu　lation，　which　is

similar　to　those　at　Gs　＝　20　kg・m’2・s－i，　as　described　above．　However，　after　attai　n　i　ng　a

maximum，　Pro　decreases　with　t，　exhibhing　a　high－ampiitude　osciliation　which　graduatly

decays　to　its　steady　value．　On　the　other　hand，　APrd　reaches　a　steady　vatue　at　a　time　（t　＝　4

ks）　much　eariier　than　P．　and　Pde．　The　signal　from　the　solid　fiow　meter　at　this　partioular　・

opera加g　cond西on　indicated廿1at　Gs　is　significantiy　affected　by　the　oscillation　of△Prd，　as

shown　in　fig．3．7．　The　ti　m　e　required　for　Gs　to　attain　a　steady　vat　ue　seems　to　ooincide　with

that　for　APrd　（t　＝　4　ks）．　Pro　and　Pde　continue　to　decrease　even　after　Gs　becomes　steady

and　fat1　off　at　pressures　about　6．2　kPa　and　zero，　respe（nively，　at　t　＝　8　ks．　Tabie　3．1

summarizes　the　steady　val　ues　of　Pro，　Pdo　and　APrd　as　well　as　the　ti　m　e，　ts，　required　for　at　I　of

廿1em　to　attain　a　steady　state．　At　G、＝10kg・m’2・S隔1，　the　steady　values　of　P（b　seem　to　be

independent　of　Hsd　and　to　att面n廿）e　steady　sねte　fa旧y　quickly（㌔＝1．0－1．5　ks）as　can　be

seen　h　Fig．3．3（a）．　At　Gs＝20　kg・m－2・s’1，　it　takes　a　longer　period　for廿1e　system　to　at圃n　a

steady　state廿1an　at　Gs＝10kg・m’2・♂．　ln　add面on，　while愉e　steady　vaiues　of　Pのare

neariy　the　same　for　all　Hsd，　Pco　and　oonsequentiy　APrd　as　well　as　ts　at　Hsd　＝　2．7　m　are

considerably　dfferent　from　those　at　Hsd　＝　32　and　4．0　m．　This　d’nterence　is　attn’buted　to　the

gas　fiow　from　the　riser　to　the　downoomer　wh　ich　results　in　a　negligible　decrease　of　the　gas

velcx　ity　in　the　riser　but　is　suracient　to　padially　fluidize　the　solid　panicles　in　the　downcomer．

On　the　other　hand，　though　the　steady　values　of　P．　and　Pde　at　G，　＝　15　kg・m’2・s’i　are

between　those　at　Gs　＝　1　o　and　20　kg・m’2・s’i，　ts　is　larger　than　those　even　at　G，　＝　20　kg・m－2・s’i　’

due　to　the　oscillating　slow　decrease　of　Pro　and　Pcb　described　i　n　Fig．3．6．
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The　ABD　distributions　varied　with　time　in　accordance　with　the　van’ations　in　P．．　The　results

are　shown　in　Fig．3．8，　where　the　ABD　distn’buti’ons　at　different　G、　are　compared　for廿1e

same　Hso　of　32　m．　As　shown　in　Figs．3．3（a）　and　（b），　P．　at　G，　＝　10　and　20　kg・m’2・s’i

reached　the　steady　vatue　so　quickiy　that　measurements　of　the　transib’onat　ABD　d’Stn’bution

were　impossible．　Thus，　in　the　figure，　only　the　steady　distn’butions　are　drawn　by　the

dashed　lines．　lt　is　ciear　from　the　dlstribut）’ons　that　at　the　former　Gs　the　whole　bed　is　in　the

state　of　dilutephase　transport　while，　at　the　latter　Gs，　dense　region　is　produced　in　the　micklle

of　the　riser．　On　the　other　hand，　the　distiibmion　at　G，　＝　15　kg・m－2・s’i　depended　on　time

duhng　the　periOd　before　P，）reached廿1e　Steady　vaiue．　ln廿〕e　figure，　two　distributions　are

representatively　drawn；　one　observed　immediately　after　G，　and　APrd　became　steady　（t　＝

42　ks）　and　th　e　oth　er　afte　r　Pro　becam　e　steady　（t　＝　82　ks）．　lt　should　be　noted　that　at　this

particu　lar　Gs，　formation　of　dense　region　takes　place　in　the　lower　part　of　th　e　riser，　as　reported

in　Chapter　2．

3．3．2　Effect　of　Downcomer－Bottom　Pressure　on　ABD　Distribution　in　Riser

　　　The　abovedescribed　effect　of　variations　in　the　pressure　at　the　riser　and　downcomer

bottoms　on　th　e　steady　state　A　B　D　distributs’on　was　furlh　er　examined　by　varyt’ng　Pcb

intentionelly　with　aeration　to　the　downoomer　bottom　at　a　rate，　ua，　ranged　from　O　to　42　×　1　06

m3・s’i．　Here，　steady　solid　ciroulation　in　the　present　system　at　aeration　rates　higherthan　42

x　1　06　m3・s’i　were　impossible　due　to　a　sudden　increase　of　Gs　as　usually　experienced　with　L－

valves　in　industねl　systems．　Figure　3．9田us廿ates　typiceti　changes　of　the　Stea（：iy－state　Pr◎

and　Pde　with　ua　at　Hsd　＝　2．7　m　for　Gs　＝　10　and　15　kg・m’2・s－i．　I　n　th　e　figure　the　dashed

horizontat　line　indicates　the　downcomer‘bottom　pressure　esti　mated　with　an　assumption

that　at1　the　particles　inhially　loaded　are　incipi’entiy　fluidized　with　a　voidage　of　O．5．　it　is　seen

from　the　figure　that　at　G，　＝　10　kg・m’2・s’i　Poo　increases　with　ua　wh　ile　Pro　remains　constant，

suggesting　that　the　ABD　distribution　is　invariable　with　ua．　On　the　other　hand，　at　Gs　＝　15

kg・m’2・s’i，　the　increase　of　u．　resutts　in　an　increase　of　Pro　as　well　as　Pcb．　Thts　cteaily

suggests　that廿1e　ABD　disthbuもon　varies　w㎞P（io　as　observed　for　the　change　of　Pd＝）w｝廿1

Hsd．　The　variation　is　confirmed　in　Fig．3．10　which　cleariy　indicates　that　the　lower　dense

region　develops　with　Pde．
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Hsd　＝　2．7　m

’
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t Fig．3．10　ABD　distribution　at　Gs＝15kg仰”2・s’1　for　various　aeration　rates
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3．4　Conclusion

　　　Measurements　were　conducted　of　the　static　pressures　at　riser　and　downoomer

bottoms　and　the　ABD　distn’butions　in　the　riser　of　an　experimenta1　CG　FB　system　which

employed　a　screw　feeder　having　a　pressure　drop　much　higher　than　widely　used　pneumatic

valves．　Within　the　present　experimented　condin’ons　the　followings　are　ooncluded：

1）　Transhional　variations　are　observed　in　廿1e　static　pressure　at　the　riser　and

　　　clowncomer　bottoms，　refleci）’ng　not　only　the　disbibution　of　solid　panicles　iniv・atly

　　　loaded　h　the　downcomer　to廿1e　hser　and　the　cbwncomer　but　also　slow

　　　developm　ent　of　the　static　pressure　distribut）’on　i　n　the　downcomer．　The　time

　　　requ　ired　to　attain　a　steady　state　is　longer　for　the　downcomer　than　for　the　riser　due　to

　　the　slow　gas　flow　from　the　n’ser　to　the　downoomer　and　the　induced　slow　gas　flow　in

　　the　upper　part　of　the　bed　in　the　ck）wnoomer　by　the　so［id　panicles　raining　down　from

　　the　cyclones　onto廿1e　bed．

2）　No　appreciable　effect　of　the　variation　of　the　downcomer　bottom　pressure，　Pde，　on　．

　　steady－state　ABD　distributions　in　a　CGFB　riser　is　found　．　Only　when　G，　takes　a

　　parj’cular　value　where　a　dense　region　appears　in　the　lower　part　of廿）e　riser，　the　riser

　　bottom　pressure　and　oorrespondingly　the　ABD　distn’buti’on　vary　with　Pde．　At　such　a

　　operating　oond謎jon，廿1e　steady」state　ABD　disbibution　depends　on　the　ini廿al　solid

　　inventory　in　the　downoomer．
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Chapter　4

HIG－VELOCIIY　FLUtDIZAnON　OF　SOLID　PARnCLES

　IN　A　CIRCULATING　LIQUID－FLUIDIZED　BED　RISER

4．1　置nt1て）du（畑on

　　　As　introduced　in　chapter　1　，　fluidization　of　solid　parddes　by　gas　always　exhibits　to　an

aggrega哲ve　fiuidiration　state・　On廿1e　o廿1er　hand，　when　fluidized　by　liquid　a　bed　of　solid

parb’des　mostiy　exhibits　partioulate　fiuidization　state　（Couderc，　1985）．　This　dnterence

between　th　e　fl　u　idization　states　was・fi　rst　examined　th　eoretically　by　Jackson　（1963）　as　a

stability　problem　of　an　incipientiy　fiuidized　bed．　He　analyzed　the　stability　based　on　a

perturbation　theory　and　evaluated　propagation　of　a　smal　loced　disturbance　in　the　solid

concentration　given　to　an　incipientiy　fluidized　bed．　As　a　result　of　the　analysis，　it　was　deatly

shown　that　growth　rate　of　the　disturbarice　in　liquidfluidizeci　beds　is　essentially　much　slower

than　that　in　gas－fluidized　beds，　i．e．　the　form　er　being　m　uch　more　stable　than　the　latter

agatnst　the　distu　rbance　and　thus　giving　a　paniculate　fluidization　state．

　　　As　described　i　n　the　precEx　Sing　chapters，　in　ci　rcu　lating　gas－fl　u　idized　bed　risers　where

solid　parl）’cles　are　fluidized　at　gas　velochies　much　higher　than　those　in　bubbling　fluidized

beds，　complicated　change　in　both　axial　and　radial　distn’but）’ons　are　observed　i　n　gas　and

solid　veloci廿es，　bed　pressure　and　voidage　not　only　w柑r　operatlng　variables　but　also　vv曲廿le

system　oonfiguration．　l　n　contrast，　a　batch　liquid‘fluidized　bed　expands　homogeneously　as

liquid　flow　rate　increases　（e．g．　Richardson　and　Zaki，　1954a）．　The　phenomena　observed　in

liquid‘fluidizec」　beds　are　much　simpler　than　those　in　gas－fluidizeci　beds．　For　example，

change　of　the　bed　voidage，　q　with　the　supedicial　liquid　velocity，　ub，　is　empirically　expressed

（Richardson　and　Zakt’，　1954b）　by

usip＝ubt（E－up．ut£rFl
（4－1）

’VVhere　u，fi，　and　up　respectively　denote　the　ljquid－solid　s　lip　velocity　and　th　e　supedicial　parl）’de

　　　　　　　　　　　　　　　　　　　　　　　　　　－61一
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velocity．　For　a　system　consisting　of　u　niform　sphericaj　partjdes，　the　exponent　n　in　Eq．4－1　is

given　as　a　fu　nct）’on　of　the　partide　Reyn　olds　n　u　m　ber　at　the　term　inat　veloclty　wh　ich　is　based

upon　th　e　diam　eter　and　the　free　fatling　velocity　of　a　single　pari）’de，　lh．　Practs’cat　appl’me’1’rty

of　the　correlation　to　particu　late　gas一　and　liquicFfl　u　idtzed　beds　has　been　expe　rimentatly

confirmed　by　many　investigators　（Garslde　and　AFDibouni，　1　977；　Rowe，　1　987；　G　ibilaro　et　at，

1　9gg）．

　　　On　the　other　hands，　Wilhelm　and　Kwauk　（1948）　first　reported　occurrence　of

aggregative　fluidization　state　in　liqu　id－fluidized　beds　oonsisting　of　high－dens’rty　padicles　like

lead　shot　and　copper　sphere．　They　observed　liquid　void　barids　moving　up　through　the

bed　and　proposed　the　parti’cle　Froude　n　u　mber　（＝　ub2・g’i・dp－i）　as　a　crtterion　for　the　transin’on

of　the　particulate　and　aggregative　fiuidization　states．　The　void　bands　were　later　confirmed

by　a　number　of　visual　observations　（Caims　and　Prausnitz，　1　961　；　Hassett，　1　961　；　Volpicelli　et

at，　1966；　Kmiec，　1978）．　Foscolo　and　Gihr’laro　（e．g．，　Foscolo　and　G　ibilaro，　1987；　G　ibilaro　et

al，　1986；　Foscolo　et　al，　1986；　Foscolo　et　al，　1984）　extensively　examined　the　transib’on

between　pa而culate　and　aggregative　fluidiration　states　based　on　an　elas廿。　wave廿1eory

（Wallis，1969）．　Based　on　the　res岨s　of廿1eir　sem●卜empirical　analysis，　they　oonstruded　a

classMcation　map　which　suggests　that　low－density　panicles　having　relatively　small　diameter

（less　than　1　mm）　aM／ays　exhibit　the　paniculate　fluidization．

　　　All　of　the　knowiedge　on　the　macroscopic　flow　characterist）’cs　described　above　have

been　oonfirm　ed　only　for　a　batch　liquid－fiuidized　bed　（BLFB）　system　where　ub　is　lower　than

the　terminal　velocity　of　a　single　parb’de，　lh．　W　hen　ub　exceeds　ut　in　B　LFB，　ali　pa市des　are

swept　away　from　the　oolumn　and　the　bed　voidage　eventuatly　reaches　unity．　Then，　’rf　the

elutn’ated　panicles　were　fed　back　to　th　e　bottom　of　the　oolumn，　as　is　usually　seen　in　a

ciroulating　gas－flujdized　bed　（CG　FB），　a　certain　value　of　the　bed　voidage　less　than　unity　can

be　obtained．　l　n　th　is　circulati　ng　liquid－fluidized　bed　（CLFB）　system，　hydrodyriam　ic

phenomena　even　in　the　liquid－fi　u　idized　bed　are　expected　to　become　complicated　because

turbuient　interadons　dominate　between廿1e　liquid　and　solid　phases．　Hence，　in　the

present　study，　measurements　were　oonducted　on　the　static　pressure　distn’bLfts’ons　through

a　nser　of　CLFB　at　various　ub　and　a’rculating　mass　flux　of　solid　particles，　Gs．　Van’ations　in

Usip　and　th　e　spatial　disbibution　of　panicles　with　ub　and　Gs　are　analyzed　to　characterize

Macroscopic　liquicS－solid　flow　structu　res　in　the　CLFB　riser．

4・2　Experi　menta　l

4・2・1　Circulating　Liquid－Fluidized　Bed　System

　　　Figure　4．1　shows　a　schematic　diagram　of　experimentat　apparatus　employed　in　the
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1，　Riser　column　（49　mm　Ld．）

2，　Slurry　Flowmeter

3，　Slurry　reservoir　（285　mm　i．d．）

4，　Water　reservoir　（285　mm　i．d．）

5，　Slurry　pump

6，　Water　pump

7，　Slurry　sampler
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Fig．4．1　Experimental　apparatus　of　circulating　liquid－solid　fluidized　bed　system
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present　study．　The　ciroulating　liquid－fiuidizecl　bed　（CLFB）　system　consists　of　a　O．049　m　i．d．

riser，　0285　m　i．d．　slurry　and　water　oolumns　and　slurry　and　water　pumps．　These　were

oonnected　with　ptpes　havi　ng　the　sarn　e　i　n　n　er　diam　eter　to　that　of　th　e　riser．　Th　e　riser　was

constructed　by　oonnecting　four　1　．0　m　high　transparent　PMMA－resin　pipes　with　one　another．

Nine　pressure　taps　were　attached　in　an　axial　distance　of　O．10　m　on　the　wall　of　each　riser　for

measurement　of　static　pressure　distributr’on．　Maximum　solid　concentration　and　feed　rate

obtatnable　by　the　slurry　pump　were　respectively　30　volO／o　and　O．4　kg・s－i．

　　　Two　differentiy　sized　glass　batlotini，　G　B93　and　G　B　I　82，　were　used　as　a　bed　materiat．

Their　size　distn’but’ons　and　relevant　properties　are　respectively　shown　in　Fig．4．2　and　Table

4．1．　According　to　the　fluidization　classification　proposed　by　Gibilaro　et　el　（G　ibilaro　et

al；1986），　as　shown　in　Fig・4・3，　the　prope面es　of廿1e　parbcies　presentjy　used　are　classMed　as

廿1e　pa！七des　which　always　exhib仕pa市culate　fluidiza廿on．

Table　4．1　Prope　rti　es　of　solid　pa　ni　cles　used

GB93 GBI82

mean　diameter，　mm

density，　kg．m’2．s’i

terminal　veloc’ity，　mm・s－i

shape

O．093

2．484

　　6．5

spherk　a1

2．380

2．480

19．9

4．22　Experimental　Proced　ure

　　　AII　experiments　were　carried　out　at　ambient　cond’nions　using　tap　water　as　a　fiuidizing

fluid．　A　weighted　amount　of　the　solid　padicles　was　loaded　in　the　slurry　reservoir（3）．　The

parb’des　are　fiuidized　by　water　at　a　velocity　required　to　matnta’n　solid　holdup　in　the　reservoir

at　a　certat　n　value　less　than　O．3，　wh　ich　cari　be　varied　by　adj　usti　ng　th　e　water　flow　rate　from

the　water　pump（6）．　Water　oveMowed　from　the　slurry　reservoir　was　led　to　the　water

reservoir（4）　and　circu　lated　by　the　water　pump．　The　solid．water　slurry　was　fed　to　a　bottom

Of　the　n’ser（1）　by　the　sl　u　rry　pump（5）．　The　ooncentration　of　the　particles　in　the　riser　was

controlled　by　adjusting　the　solid　holdup　in　the　slumゾreservoir　and　dilし市ng廿1e　slurTy　feed　vv㎞

Water　fed　by　the　water　pump　immediately　below　the　slurry　flow　meter（2）．　All　effluent　from

　　　　　　　　　　　　　　　　　　　　　　　　　　－an一
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Fig．4．3　Gibilaro　et　al’s　criterion　for　fluidization　regimes
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廿le　top　of廿）e　riser　was　led　to廿1e　sluny　reservoir　or　to　a　slurry　sampler（7）．　The　operating

variades　such　as　ub　and　Gs　as　well　as　the　steady　state　of　the　whole　system　were

confirmed　by　both　the　static　pressure　distribmion　i　n　th　e　riser　and　the　slurry　sampling　where

weighi，　Ws，　and　volume，　Vs，　sampled　for　a　certain　sampling　time，　lmp，　were　measured．

The　sarnpling　was　repeated　more　than　4　times　for　each　run．　The　static　pressures　at

dnterent　heights　along　the　riser　were　m　easured　by　a　pressure　transducer　via．　solenoid

vatves　wh　ich　were　actuated　by　a　personal　computer．

4．23　Circulati　ng　Opetation

　　　The　solid　ooncentration　in　the　slurry，　ps，　the　voidage，　Es，　as　well　as　ub　and　Gs　were

determined　from　the　slurry　sarnpling　as　follows；

　　　　p，＝W，IV，＝pt（1　一q）＋p・q　（42）

　　　　q＝　op　一一　p，）／（pp－p）　（4－3）
　　　　Gs＝　Pp　Vs　（1　“Es）／（Ar’tdamp）　（‘va）　．

　　　　ub＝Vs　q／　（Arim）　（4s）

where　Ar　denotes　the　cross　sectionat　area　of　the　riser　column．　The　mean　bed　voldage　in

the　riser，　E，　was　calculated　from　th　e　slope　of　a　static　pressure　distribution　cu　rve．　A

negligibie　oonbibuts’on　of　the　wat　l　fricb’on　to　the　disbibution　was　detected，　wh　ich　was

eveluated　to　be　less　than　5　O／o　of　the　measured　pressure　at　370　mm・s－i，　i．e．，　the　maxi’mum

supenia’al　liquid　velocity　in　the　present　experiment　From　Eqs．　4－1　to　4－5，　the　liquid－solid

slip　velo⊂叙in廿le　riser，　us晶P，　is　expressed　as；

Usfip　＝　Vs　｛Es　／E一　（1　’一　Es）　／　（1　一　E）｝！　（Ar’tsarnp）
（妬）

Since　no　downflow　of　the　parb’cles　near　the　riser　wall　was　observed　at　ub　higher　than　lh，

thus，　the　following　relationship　can　be　obtained　from　Eq．4－6：

Es＞E （4，D

The　above　oondta’on　was　employed　to　check　if　the　slurry　sampling　was　correctiy　canied　out

or　not

4・3　Results　and　Discussion
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4．3．1　Bed　Expa　nsion　Cha　racteristi　cs

　　　At　superf“ct’al　ltaukl　velocities，　ub，　lower　than　the　terminel　velocity，　lh，　the　beds　exhibited

pa面culate価diza虹on　for　both　GB93　and　GB182　pa応des　as　expeded　from　Foscolo　et　al’s

criterion　（Foscolo　et　al，　1984）．　I　n　Fi　g．4．4　th　e　bed　voidage，　E，　is　plotted　against　ub　on　a

｝ogarith　mic　scale．　This　result　was　obtained　from　a　preliminary　experiment　by　using　a

batch　liquid‘fiu’k　iized　bed　（BLFB）　system．　For　both　partides，　th　e　relationship　between　E

and　ub　can　be　well　represented　by　a　straight　line．　lt　is　well　known　that　the　relationship　can

be　practically　described　by　廿le　so＜＞alled　Richardson，Zaki　（R－Z）　equation，　Eq．4－

1（Richardson　and　Zaki，　1954a）．　The　index　n　for　E　in　Eq．4－1　for　each　particle　bed　is

respectively　found　to　be　4．5　and　3．8　from　the　linear　relationship．

　　　At　ub　h　igher　than　ut　with　cirou　lation　of　th　e　elutn’ated　partides，　a　certain　voidage　was

obt釦ned廿1roughout廿1e　riser．　Figure　4．5　shows　typica　I　axia　I　distribし紺ons　of廿le　static

pressure　in　the　riser．　The　disin’buts’ons　were　obtained　at　ub　ranged　from　141　to　1　68　m　m・s－i

for　GB182．　lt　is　cleariy　seen廿1at廿1e　static　pressure　decreases　lineariy　with　the　height

廿iroughoしrt廿1e　riser・　　lt　should　be　noted　that　the　so一（＞alled　S－shaped　distribし由on　which　is、

frequentiy　observed　in　ca　FB　risers　is　never　produced　wtth　in　the　present　operating

condin’ons．　I　n　ack　idion，　descent　of　panicles　near　the　riser　wedl　was　never　observed，　which

is　usuatly　observed　in　both　CGFB　and　BLFB．　Figure　4．6　shows　typical　variations　of　mean

bed　voidage，ε，　with　Gs　for　vari●ous　ub．　lt　can　be　seen廿）at　E　increases　with　Gs．　The　rate

of　decrease　of　E　is　fi　rst　rapid　and　then　g　radually　deteriorates　as　Gs　increases．　Furtherm　ore，

it　should　be　noted　that　a　sign面cant　increase　ofεwith　increasing　of　ub．　ln　par廿cular，　for廿1e

coarser　padicles，　an　apparent　increase　in　E　can　be　seen　as　ub　is　varied　from　23r29　mm・s’i

to　151一一181　m　m・s’”，　wh　ich　might　be　brought　about　by　a　transin’on　of　flow　state　from

paniou　late　fl　u　idization　to　agg　regative．

　　　4．3．2　Liquid－Solid　Slip　Velocity

　　　The　observed　E　at　d’nterent　ub　is　plotted　against　usfip　ceilou　lated　from　Eq．4－6　for　GB　92

and　GB　182　in　Fig．4・7．　Here，廿1e　resu比s　obtained　in　BLFB　are　aiso　plotted　in廿le　figure．

lt　is　cteardy　seen　that　usip　in　CLFB　becomes　oonsiderably　higher　than　that　in　BLFB　for　both

partide　beds．　A　sharp　and　linear　increase　of　E　with　usfip　can　be　seen　at　lower　ub　of　80－95

mm・s－i　for　GB93　and　23－29　mm・s’i　for　G　B　182　respedwely，　as　is　seen　in　BLFB．　On　the

Other　hand，　the　increase　of　E　deteriorates　gradually　as　ub　becomes　higher．　Comparing

the　existing　oorrelation　as　drawn　by　the　sold　line，　R－Z　equation　is　found　to　be　no　more

apPlicable　to　describe　the　relationship　betweenεand　u鋲）in　CLFB　riser．　Since廿le

fiuidizing　water　fiows　through　intersti’ce　among　the　pardcles　the　exist）’ng　equations　for　fiow

　　　　　　　　　　　　　　　　　　　　　　　　　－68一
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Fig．4．5　Typical　distributions　of　static　pressure　in　CLFB　riser　for　GBI82
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Fig．4．6　Variation　of　bed　voidage　with　Gs　at　various　ulo　in　CLFB　riser

　　　　for　different　particles；　（a）　GB93　and　（b）GBI82．
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through　a　packed　bed　（Bird　et　el，　1　960）　cayi　rationally　be　applied　for　those　data：

u，ip　＝　（　p，　一　p　）・g・cb2・if／（36・K・（1　一s　））
（4－8）

Here，　K　denotes　tortuosity　of　liquid　flow　through　the　bed　and　the　foliowing　values　were

employed　from　the　welH〈nown　equations：

K＝　2　；　the　modMed　Hagen　一　Poiseulle　equation

K＝　25／6　；　the　Blake　一　Kozeny　equation

These　equa哲ons　are　also　drawn　in廿）e　figures　by　broken　and　dot－dash　lines　and　oompared

with　the　data．　From　the　figures，　an　obvious　deviation　between　the　observed　data　and　the

ptedicted　results　is　found，　a　quat　itative　agreement　between　them　can　be　seen　at　h　igher　ub

for　both　panicle　beds　though．　This　discrepancy　may　be　attributed　to　the　unsteady　and

complex　fiow　structures　in　the　CLFB　as　discussed　below．

4．3．3　Transiv’on　of　Fluidization　State

　　　Figure　4．8　shows　typicat　photographs　taken　through　the　riser　wall　at　a　height　of　1　．8　m

from　the　riser　bottom　for　G　B　I　82．　Here，　the　solid　flow　in　the　riser　was　visuallzed　by　back－

lighting　through　the　bed．　Both　photographs　were　taken　at　condta’ons　where　the　bed

voidages　were　approximately　the　sam　e　but　ub　is　considerably　different　l　n　the　photograph

（a）　taken　at　ub　＝　101　mm・s’i，　a　homogeneous　shadow　can　be　seen　over　the　whole　riser．

This　suggests　that　th　e　bed　is　i　n　the　parljcu　lately　fl　u　idized　state　as　predicted　by　Foscolo　and

Gibilaro　（Foscolo　and　Gibilaro，　1984）　while　the　bed　at　ub　＝　371　mm・s’i　is　no　more

pa面culate　as　can　be　seen　in　the　photograph（b）．　At廿1e　lower　ub，廿1e　partjcles　were

transported　upward　in　a　laminar　manner．　At　the　higher　ub，　on　the　other　hand，　the　parb’des

tend　to　form　aggregates　haVing　irregular　shapes　in　the　riser　and　the価dizing　water　passes

among　the　aggregates　in　the　form　of　streaks．　Thus，　complex　motions　of　padicles　and

inhomogeneous　disbibuin’on　of　the　bed　voidage　were　eventUa」ly　observed　through廿1e

CLFB　riser　wall　as　suggested　elsewhere　（Grbavc：ic　et　al，　1　992；　liang　et　al，　1　996）．

　　　The　inhomogene’rty　’in　the　spatial　distn’bmion　of　the　solid　parts’cles　is　further　examined

by　anatyzing　the　luminous　intensity　disbibution　in　the　photographic　images　through　the

fOllovving　procedure．　First，　the　photograph　ic　image　was　oonverted　to　a　bit　map　image

consisting　of　256　gray　sceties　by　digin’z’ng　spatial　disbl’butjon　of　the　luminous　intensity　over

the　photograph　with　an　image　scanner．　The　digittz’　ed　data　were　stored　in　a　personat　・

　　　　　　　　　　　　　　　　　　　　　　　　　　一73一
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（a）　E一　o．goo （b）　E＝　O．895

Fig・4．8　Typical　photographs　of　backlighted　bed　of　GBI82　taken　at　different　operating

　　　conditions；　（a）　ui，　＝　101　mms－i，　G，　＝　26．8　kgtm－2・s’i　and　（b）　ul，　＝　371　mm・s－i

　　　Gs　＝　111．1　kg－m－2・s’1
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computer．　Then，　the　luminous　intensity　at　each　pixel，　which　corresponds　to　the　parb’at　bed

voidage，　was　transverseiy　scan　ned　along　certain　horizontad　l　ines　chosen　anditran’ 撃凵@on　the

photograph．　Typical　raw　signat　for　different　ub　obtained　from　th　e　above　procedure　is

shown　in　fig．4．9．　Here，　since　base　level　of　transm　ittancy　through　the　riser　is　noticeai　”y

distorted　due　to　the　geometry　of　the　riser　oolumn，　a　quadratic　regression　curve　was

analyti’cel　ly　established　to　frt　raw　signal　with　oorrelation　coethcients　above　O．93．　The

distonion　was　equalized　by　subtract）’ng　the　data　from　the　regression　cu　rve．　Then

successive　deviations　of　the　data　from　the　regression　cu　rve　can　be　extracted，　as　drawn　in

廿1e　same　figure・　　lt　is　cleaudy　found廿1at廿1e　spa廿al　flu（カ」a哲on　obtained　at　the　higher　ub

（fig．4．8fo））　can　be　characterized　as　a　signal　exhibiv’ng　large　amplitude　and　sharp　spikes，

compared　to　that　obtained　at　the　Iower　ub　（Fig．4．8（a））．　The　fluctuations　of　the　l　uminous

intensity　for　different　ub　were　anEdyzed　by　th　e　power　spectrEd　density　（PSD）　function．　I　n

Figs．4．10　and　4．11，　typical　PS　D　fu　nctl’ons　calculated　from　the　extracted　fluctuations

obta’ned　for　dnterent　ulo　are　respectively　drawn　on　normEd　（Fig．4．10）　and・logarithmic

（Hg．1．11）　diagrams．　Here，　sampling　window　used　for　anatyzing　the　PSD　contatns　4096

POints・　in　bo廿1　figures，廿1e　absdssa　denotes　the　spatia　l　frequency．　As　can　be　seen　in

Fig．4．10，　several　peaks　appear　with　in　th　e　range　from　O．2　to　O．5　m　m－i　for　higher　ub　while

such　peaks　become　ambiguous　for　lower　ub．　The　result　suggests　that　the　spatiat　scade　of

the　aggregates　observed　through　the　riser　can　be　esb’mated　ranged　from　2　to　5　mm．　ln

addin’on，　Hg．4．1　1　shows　that　the　PSD　fu　ncb’on　exhibits　a　continuous　spectrum　with　many

spikes　and　expOnential　decay　h　the　higher　frequency　region　for　bo廿1　extreme　fluidization

states．　The　appearance　of　the　PS　D　function　is　generally　att）1’buted　to　chaotic　fluctuation　of

the　signat　ooncemed　and　may　refiect　the　fractal　textu　re　in　the　spatial　distn’bmion　of　the

voidage　through廿1e　riser．

　　　Thus，　diagnosis　on　the　oomplexity　of　the　luminous　intensity　fluctuation　for　dnterent

fluidization　regimes　was　oonducted　on　the　basis　of　a　fractEd　anatysis．　Fractal　dimension，　Df，

has　extensively　been　used　as　a　quantifier　of　geomebic　complexity　of　an　object　（Feder，

1988；　Hilbom，　1994；　Luewisuthichat　et　al，　1996）．　The　dimension　is　practjcally　calculated

through　the　box－counting　method　which　is　employed　in　the　present　work．　The　fl　uctuati　ng

Signel　of　the　luminous　intensity　was　mapped　onto　a　regulai’　rriesh　of　size，　s，　as　shown　in

Fig・4・12，　and廿1en廿1e　number　of　boxes　required　to　cover　the　cun！e，　N（s＞，　was　oounted．

Vatiations　in　N（s）　with　progressively　decreasing　s　were　plotted　on　a　logarithmic　diagram．

On　the　diagram　the　linear　relationship　between　N（s）　and　s　is　obtati’ned　and　the　slope　directly

yields　the　fracted　dimension，　Df．　Figure　4．13　shows　the　typical　change　in　number　of　the

boxes　with　the　segment　size　fro　m　1　O　to　1　OO　pixels　wh　ich　corresponds　to　th　e　length　from
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o．15　to　2．93　mm　on　the　originat　photography．　As　can　be　seen　in　the　figure，　the　linear

relationships　between　N（s）　and　s　are　obtatined　for　both　particu　late　fl　u　icltzation　and

aggregative　one　and　provide　noninteger　dimensions．　Averaged　Df　for　the　parb’oulate　and

aggregative　fiuidizations　is　respectjvely　found　as　1　．45　and　1　．39．　The　slightiy　lower　Df　in　the

aggregative　fiuidization　state　indicates　that　the　degree　of　complexity　of　the　voidage

disnibuti・on　in　the　riser　decreases　due　to　the　formation　of　the　larger　scale　aggregates．

4，4　Conciusion

　　　Systematic　measurements　were　canied　out　of　pressure　disbibuti’on　through　a

circulating　liquid－fluidized　bed　（CLFB）　riser　at　various　ub　and　G，．　The　fiow　structu　re　of　solid

partides　was　experimentally　examined　for　dnterent　sized　particles．　Within　the　present

experimental　oondta’ons，　the　follovvings　are　ooncluded：

1）　・　A＞cially　homogeneous　distribut’ons　of　the　solid　hokdup　are　abvvay＄，　obsetved　in　CLFB

　　　riser．　　The　s（〉一（）alled　S－shaρed　distribし面on，　which　is　frequentiy　observed　in

　　　drculatjng　gas　fluidized　bed　risers，　is　never　produced　in廿1e　CLFB　hser．

2）　Exist）’ng　flow　models　obtatned　for　parljculate　beds　or　packed　beds　become

　　　inapρiicabie　to　describe廿le　relationship　between　the　liquid－solid　slip　velock〆and廿le

　　　bed　voidage　in　CLFB．

3）　Bed　which　always　exhibits　particulate　fluidization　at　ub　less　than　term　inal　velocity，　ut，

　　　tenctss　to　form　aggregates　in　the　CLFB　riser　at　ub　〉　ut．　At　this　oondta’on，　a　fiuidizing

　　　liquid　eventually　passes　th　rough　the　bed　in　the　state　of　streak．

4）　Spa廿al　dis面bしnion　of　the　bed　voidage　in　CLFB　riser　exhibits　fractal　stn」ctu　re．　Lower

　　　fractal　dimension　is　obtatned　at　ub　appreciably　h　igher　than　ut，　which　is　attributed　to

　　the　formation　of　larger　sca］e　aggregates．
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Chapter　5

CHAOIIC　CHARACTERISnCS　OF　LIQUID－SOLID　FLOW

　IN　A　CIRCULATING　LIQUID－FLUIDIZED　BED　RISER

5．1　introcl　uction

　　　ln　the　previous　chapter，　even　liqu　ict－fi　ukl’ved　solkl　partides　exh　ibit　an　aggregative

fiuidiZation　sねte　when廿le　pa面des　are　fiuirdtzed　at　superfb創liquid　vei（x）耐es，　ub，　higher廿1an

the　terminel　veloclty　of　the　single　nve，　lh．　The　state　has　never　been　observed　in　exisb’ng

investl’№≠狽奄盾獅刀@since　atI　experimentS　were　cam’ed　oUt　in　batch価dized　bed　at　ub＜ut．　hthe

state，　fiuctuations　with　time　are　observed　in　most　of　flow　characteristk’　）s　such　as　the　solid　and

liquid　vei（）cin’es，廿）e　statk）bed　pressu　re　and廿1　e　bcal　bed　voidage（e．9．　Luewisし曲bhat　et　al，

1995）・　These　flu（：加ations　vvou［d　refi∈℃t　turbulent　hydrodynamb　htera（：カons　betw∈×∋n廿le

sdid　particrles　and　1igu’ri　which　may　be　induced　by　the　high　siip　velocity　between　them．

　　ln　the　fiekl　of　chem’k　Eal　reactor　engineering，　it　has　been　made　dear　that　muhiphase

reactors　frequendy　exh　ibit　Chaotic　fiu（カ」ations　of廿le　above　vekx蒲es，　pressure，　vo’ktage　etc，

as　a　direct　consequence　of　the　tu　rbu　lent　interactions　between　the　phases．　Recendy，

deterministil）　chaos　theory　has　successfu　lly　been　ack）pted　to　anatyze　oゆUt　sig　nalS　from

van’盾浮刀@probes　for　the　complex　phenomena　（Hilbon；　1994）．　On　the　ms’s　of　the　theory，

complexity　of　turbulent　phenomena　is　quantitied　in　terms　of　the　correlation　dimension　arrci　the

Kolmogorov　entroP》へ　In　most　of　exis百ng　inves噸bons，廿1e　chaotk）fbw　strudure　of

muhiphase　system　has　been　characterized　by　pressure　fl　uctuations　in　the　bed　（e．g．　Bleek　and

Schouten；　1　993，　Bouillard　and　Miller，　1　994，　Daw　et　al；　1　990，　Huilin　et　at；　1　997）．

　　Variation　of　the　fluklization　state　with　ligukl　velocity　would　result　in　that　of　heat　and　mass

transfer　characteristk）s　in　the　bed．　Usuatly，　time－averaged　heat　transfer　characteristics　have

been　invesit）’9ated　at　various　operating　condhlons　in　batch　iiquic団uたiized　beds　for廿lose

between　the　bed　and　oolumn　watl　（Fan；　1　989，　Kato　et　Eal；　1981　，　Richardson　and　Mitson；　1958，

Richardson　and　Smith；　1962）　or　for　those　betvveen　the　bed　and　immersed　oicijects

（Richardson　et　al；1976，　Khan　et　al；1983）．　These　investigations　provide　an　insight　into廿1e

Steady　heat　transfer　characterisb’cs　but　never　seem　to　be　suraa’ent　to　expleln　instantaneous

Characteristi’cs　which　strongly　depend　on　the　fluctuations　in　liqu　id－solid　fiow　m　entioned　above．
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　　ln　th　is　chapter，　the　detemiinist）’c　chaos　analysis　has　been　applied　to　time－series　data　of

k）cel　voidage　arKsi　heat　transfer　fiuctuations　measured　by　novel　probes　deveicped　here　for

exFManation　of　thetr　variations　with　operating　condin’on　as　well　as　their　corresporK　ience　to

those　of　flow　pattems　in　a　a’rculating　liquidfi　u’dtzed　bed　riser．

5．2　Expe補階rnenta　l

5．21　Probes

　　　figure　5．1　shows　a　schematic　diagram　of　the　experimentEal　circulating　1jou’xt‘fluidized

bed　system．　（）perating　procedure　of　the　system　has　at　ready　been　descn’bed　in　the

ptececfing　chapter　in　detatl．　l　n　order　to　exam　ine　van’ation　of　the　local　flow　pattem　of　the　solid

parti’des　and　heat　transfer　rate　wtth　time，　an　opticEal　transmittance　probe　and　heat　transfer

ptobe（8）　have　been　employed．　Measurement　system　and　the　deta’ls　of　the　probes

ptesentiy　developed　are　shown　in　fig．　52（a）　＆　（b），　respectively．　The　icx　el　voiclage　was

detected　with　the　opticetl　transmitlance　probe．　The　probe　consists　of　light　emim　ng　tube　and

detecti’ng　opticeil　fibers　supported　by　4．0　mm　o．d．　stEtinless　tube　（fig．52（a））．　An　atuminum

plate　with　a　pt’nhole　was　attached　at　the　tip　of　the　emrajng　tube　to　narrow　the　He－Ne　laser

beam　with　in’niel　diarneter　of　2．0　mm　to　02　mm，　wh℃h　enabie　to　detect　a　passage　of　a　single

parb’de　（Kkuch　i　et　at；　1996）．　The　narrowed　beam　introduced　through　the　pinhole　was

received　by　the　detecb’ng　optjcel　fiber　having　a　diameter　of　O．5　mm，　whk　h　was　faced　to　the

emim’ng　fiber　tip　with　a　gap　of　4．0　mm．　The　bearn　transmnted　through　the　bed　was

introduced　to廿1e　photo廿ansistor　via・the　oP廿cal　fiber　and　converted　into廿1e　vohage　oし劇L

On　the　other　hand，　the　instantaneous　heat　tmmster　rate　is　measured　with　a　heat　transfer

ptobe．　The　probe　is　developed　based　on　the　princtple　of　the　constant－temperature　hotIwi’re

anemometry．　A　thin　piece　of　tu　ngsten　vvire　having　a　diameter　of　10　1．mi　was　employed　as　a

detecting　filament　The　wire　was　stretched　between　two　tips　of　stEti’nless　supports　and

welded　with　a　spot　wek　ier．　A　gap　between　the　supports　is　exactiy　fixed　to　2．0　mm．　The

wt　re　and　the　supports　were　coated　with　aii　anti－water　vamish　to　prevent　a　direct　oontact　of

Water　with　the　filament　Then，　the　supports　were　oonnected　to　one　of　the　arms　in　the

Wheatstone’s　bridge　circuit　A　block　diagram　of　the　electrlc　bridge　circuit　presentiy　used　is

　　The　optk）ad　fiber　probe　was　horizontatily　mounted　at　a　riser　height　of　3．1　m　and　the

emitb’ng　and　detecb’ng　tips　were　located　at　the　axis　of　the　riser．　The　heat　tiansfer　probe　was

aso　iriserted　hortzont副ly　at廿）e　same　Ievel　of廿1e　riser　and　the　tip　was　located　very　dose　to

that　of廿1e　oF棚probe，　as　shown　in　Rg．52（b）．
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1　，　Riser　column　（49　mm　i．d．）

2，　Slurry　Flowmeter

3，　Slurry　reservoir　（285　mm　i．d．）

4，　Water　reservoir　（285　mm　i．d．）

5，　Slurry　pump

6，　Water　pump

7，　Slurry　sampler

8，　Optical　＆　Heat　transfer　probes
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Fig．5．1　Experimental　apparatus　of　circulating　liquid－fluidized　bed　system．
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Fig．5．3　Schematic　diagram　of　electric　circuit　for　constant　temperature　anemometry
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　　The　signals　obtatned　by　both　probes　were　recorded　with　a　data　recorder．　The　data

was　dig｝泣zed　w曲A／D　converter　at　a　sampling　inter＞al　of　1／1000　sec　ar】d　stored　as　a

successive　data　in　a　personat　computer．

522　Data　Analysis

　　　The　dynamic　flow　characteristics　in　the　a’rculating　lquid‘fluidved　bed　riser　is　represented

with　an　attractor．　The　attractor　is　practicatly　reconstructed　in　the　m　uboimensionat　state

space　by　using　the　time　series　data　of　the　system’s　van’aiotes　such　as　pressure，　voidage，

velocity　and　so　on　obtat’ned　by　the　spedfk　method．　l　n　the　present　study，　the　attractor　has

been　reconstructed　from　the　ti　me　series　data　of　the　1　ight　transmittance　and　heat　transfer　rate

based　on　Takens’　embedding　theorem　（Takens；　1981）．　He　proved　that　the　geometric

oomplexjty　of　the　attractor　reconstructed　in　hyper　state　space　from　the　one－dimensionat　time

series　data　becomes　equivelent　to　that　of　the　actuat　system　behavior　when　embedding

dimension　attat’ns　twice　of　the　degree　of　the　freedom　oonta’ned　in　the　system　concemed．

Figu陀5．4　de岡s　the　concept　of　the　reconstruc百on　of廿1e　aU旧dor　h　3－dimensbnd　sセite

space　from　the　sucx）essive　time－series　data．　Usuatly，　the　data　is　successively　measured

and　recorded　at　a　certain　time　intervat，　T，　as　follow7．

柳・痢λ鳩λ…脚・rt4一＋τ）・痢＋2τλ…痢＋‘η一1／o，鳩＋ητ♪

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　づFrom　the　measured　time　series　data，　the（kdimensionat　state　vector，　Xi　，　IS　reconstnucteci　as

fdlowhg：

づ
X’一｛xω・x（ti＋τ）・x（ti＋2τ）・…・x（ti＋（d一一2）τ），x（ti＋（d一一1）τ）｝（51）

where　x（e　and　d　denote　the　successive　data　of　voiciage　or　heat　transfer　and　the　embedding

dimension，　respectively．　Generatly，　an　orbit　of　a　series　of　the　above　state　vector　becomes

geometriceil　ly　simpte　when　the　flow　in　th　e　riser　predominantiy　indicates　a　petiod℃　behavior．　lf

the　system　exh　ibits　chaotic　fl　uctuation，　the　geometric　structure　of　the　attractor　becomes

inevitably　complex．　The　complexity　of　the　reconstructed　attractor，　whic）h　corresponds　to　the

degree　of　the　freedom　in　the　attEti’nat　ge　states　of　the　system，　is　anaiytjcedly　evel　uated　with　the

Chaos　statisti’cs　such　as　correlation　dimension　and　Kolmogorov　entropy　of　the　reconstructed

attraCtor廿1　rough廿ie　follovVing　Procedure．
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　　　ln　the　pc’esent　study，　the　correlation　dimension，　D，，　whioh　can　be　interpreted　as　a　fractat

dimerw’on　of　the　orbit　of　the　attractor　in　the　（　kiimensional　state　space，　is　detemiined　through

伽e創9・舳m騨byG触◎α網P㎜ccb（1983）．　F・r　each　point　i。f崩e醐。f

the　oecimensionat　attractor，　the　number　of　the　points　of　the　orbit　Mng　within　the　distance　rfrom

the　point　iexctuding　the　point　i’rtseff　is　counted　（fig．　5．5）．　Here，　correlation　integrat，　C（D，　is

由剛t・be　the　relatiVe　number・f崩e　p・inお、舳in出ed曲ce　r麟ef・il・陶閃ua石。n：

c（・）＝ 香i“｝一1）甑象ノ、◎｛・一区刈 （52）

where　m　is　totat　number　of　reconstructed　vectors　consisb’ng　the　attractor，　r　is　the　radius　of　a

hyper　sphere　centered　on廿1e　point　i　in廿le　state　space　and《D（z）k3　the　Heavisk＝k∋funcb’on

（〈E）（z）　＝　1　for　z　〉　O，　（D（z）　＝　O　for　z　〈　O）．　For　determ　inisticaily　chaotic　systems，　the　correlation

integral　obE）ys廿）e　pOvver　law　over　a　sign桁cant　m　nge　of　r．

C（r）◎くrD・
（5－3）

and　the　slope　of　the　linear　part　on　a　）og（r＞　log（C（r））　ptots　yields　the　oorrelation　dimension，　Do

　　On　the　other　hand，　Schouten　et　at．　（Schouten　et　at；　1994a，　Schouten；　1994b）

introduced　the　Kolmogorov　entropy，　K，　of　the　reconstructed　attractor　for　est）’mation　of　th　e

entropy　of　the　attat’nable　states　of　the　mula’垂?ａｓｅ　system　based　on　the　maximum－likelihood

approach．　ln廿iis　EdgOrithm，　the　entropy　was　calculated　from　the　average　number　of　steps

requisite　for　a　pair　of　vectors，　which　are　inhi’ally　within　a　specific　ma）q’mum　length，　ts，　to

separate　until廿1e　distance　be瞭∋en廿）e　pair　is　larger廿1an　lc．　For　all廿1e　pairs　of　POints　on　an

atuactor，　the　number　of　steps　was　mmlated　and　then　the　average　number　of　steps　was

estimated．　h廿lis　waY，　the　Kolmogorov　entropy　was　defined　as　follovvs：

K　＝　一log，　（1　一　，lig一）

（5－4）

where　b一@’is　the　average　number　of　steps　for　which　the　d’Stance　becomes　largerthan　le．

5・3　Resutts　and　Discussion

5・3．t　Variatl’on　of　Local　Bed　Voi’dage　and　HeatTransfer

　　figuies　5．6　＆　5．7　show　tyl　X’cEal　time　series　output　signats　simultaneously　obtEtined　with

the　opb’Cetl　transrn　ittance　probe　and　the　heat　transier　probe　for　a　sequence　of　a’rculating　solid

Mass　fiux　at　a　ligukl　velocity　of　O．1　07　m・s’i　．　Without　the　circulation　of　solld　parljcles
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（Fig．5．6（a）），　the　output　vottage　of　the　t］ansm　ittarK　xg　probe　is　el　most　constant　e．g．　1　．8　V．

Here，　sudden　drcps　h廿1e　ou馴v（：＞ttage　are　frequen廿y　seen　i　n廿1e　figure　because　small

arnou　nt　of　parbdes　vvas　oontained　in　the　water　and　a　nVe　i　nter（：印ts廿le　naπowed　laser

beam　．　As　increasing　Gs，　the　bed　vcidage　correspondingly　decreases　and　everrtuel　ly　time

occupled　by　lnu’rd　phase　in　the　measuring　region　’s　seen　to　decrease．　At　Gs　larger　than

4g．18　kg・m2・s’i，　the　beam　is　intercepted　by　solkl　phase　during　most　of　the　measuring　time．

ThiS　van’ation　of　the　opm劇　signal　with　Gs　deaiiy　ind’K）ates廿1at廿ie　solid　phase　graduatly

tans就s　frて）m　the　dispersed　phase　to　con廿nuous　phase　as　a　resu　lt　of　th　e　increase　in　Gs．

Fgure　5．7　shows　oulput　signats　from　the　heat　transfer　probe，　simultaneously　obtatned　at

van●盾浮刀@Gs　corresponding　to廿1e　opera廿ng　oond市ons　in　Fig．5．6．．　tt　shoukl　be　noted　that　a

largescale　periodic　osa’llation　i　n　the　output　voltage　for　the　heat　transfer　probe　ts　seen　in　Fig．

5．7．　This　is　due　to　the　osa’llation　in　the　solkS－water　slurry　feed　induced　by　the　slurry　pump

whicli　was　seen　only　at　lqu’ri　velocities　lower　thari　about　O．1　m・sl　lt　is　found　from　the　figure

that　the　timeaveraged　output　voltage　seems　to　be　constant　in　the　range　of　Gs　lower　than

49．1　8　kg・m2・s－i　．　（）n　the　other　hand，　when　the　sol’ri　phase　becomes　continuous　phase　in

the　riser　at　G，　＝　64．90　kg・m’2・s’i，　the　output　signat　for　the　heat　transfer　obviously　increases．

From　these　van’ations，　it　can　be　oonsidered　that　the　heat　transfer　may　be　enhanced　due　to

廿1e　decrease　in　voi（ねge　because　local　tu　rt）ulence　is　inevit］∋by　induced　by　the　fi　u’rdized

nves　around　th　e　probe　tip．　Egures　5．8　＆　5．9　shows　the　output　signats　from　the　opticat

and　heat　transfer　probes　for　dMerent　G、　at　ub＝O．320　m・s’1．　The　vaね廿on　of　the　optた創

signats　with　Gs　at　high　uio　seems　to　be　similar　to　that　at　ub　＝　O．1　07　m・s－i．　lt　is　seen　that　the

voidage　fiuctuates　vigorously　with　time，　exhibhing　spiketike　increases．　This　is　atl）lbuセわle　to

廿1e　Iiquid　sb・eaks　whk：ナ1　have　been　observed　at　h　igher　ub　as　described　in廿l　e　preceding

chapter．　lt　is　deariy　seen　that　the　inorease　in　ub　from　O．1　07　m・s－i　to　O．320　m・s一‘　gives　rise　to

Increase　l　n廿）e　oし咋）Ut　votage　h廿1e　heat　transfer　prて丈）e　in廿1e　bed　and　to　vigorous　flu（：カUations

with　time．

　　From　these　ti　m　e　series　data　from　the　’heat　transfer　probe，　variation　of　the　time－averaged

heatセansfer　coe網ent　h，’s　obセユlned　by廿1e　folIowi㎎equation二

h＝

Vp　2・R，

n・d．．　・1，，　・AT・（R，　＋　R，　）2

（55）

where　R　iS　the　wire　resistanoe，　Rb　is崩e　resis臨e　of　the　bridge　of　the　Wheatstone’s　bridge　in

Series　with　Rt　and　AT　is　the　dMerence　in　temperature　between　the　vvire　and　the　bu　lk　of　the

bed，　TQ　Rt　is　di　rectiy　ca℃u　lated　from　the　vvi　re　tempe　ratu　re　by　the　following　equation：

　　　　　　　　　　　　　　　　　　　　　　一96一
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Rt　＝po’　i．；El－s一’i”，　［i＋o・（T－T．）］

（56）

For　tungsten　vvire，　thermed　coefficient，　u，　and　electric　conductMty，　po，　are　5．3　x　10B　Ki　and

s．35　x　1　08　W・m　at　300　K，　respect」vely．　Rt　is　fixed　once　the　wire　temperature　is　set

　　　ObtEti’ned　time－averaged　heat　transfer　coeffwhent　h，　ts　plotted　as　a　functjon　of　bed

voidage　at　durerent　ub　in　Fig．5．10．　At　ub　＝　O．107　m・s’‘　and　O．139　m・s’i，　h　increases　when

the　bed　voidage　becomes　lower　than　around　O．82　wh　ile　no　significant　dependency　on　the

vddage　can　be　seen　in廿ie　mmge　of　voidage　larg〈∋r廿1an　O．85．　Th　is　increase　k3　atthbu圃e　to

the　promotion　of　the　k　x　ad　tu　toulence　and　addhionat　surface　renewEal　in　th　e　thermat　boundary

layer　at　the　wire　surface．　Further　increase　in　the　ljould　velocity　up　to　ub　＝　O．320　m・s－i　gives

rise　to　a　remarkable　increases　in　h．　lt　is　atso　seen　that　h　is　fou　nd　to　increase　with　decrease

in廿le　bed　vo’k　lage・　ln　Fig・5・11the　heat　tranSfer　coeffbient　oヒ」tiヨl　ned　at　d膏ferent　ub　b　pbtted

as　a　function　of　uso　Here，　heat　transfer　coeffxx’ent　in　a　batch　liqu　id　fl　u　kltzed　bed，　wh　ich　was

obtaineXi　th　rて）ugh　the　hot→vVire　anemometry　tech　n　igue　by　Boerefijn　et…al（1996），　is　dso囹。廿ed

and　compared　with　the　present　results．　I　n　compan’son　with　the　resu　lt　obtatned　in　the　batch

liquid　fiuidizeci　bed，　higher　h　is　obtatned　in　CLFB　riser．

5．3．2　Chaotic　AniEi　ty　sis

　　　Accordhg　to廿1e　above　findings，廿le　ioca1　hear　transfer　sセて）ngly　depends　on　bed

vcidage　and　ub，　which　can　be　considered　as　the　resutt　of　the　change　of　the　iocal　turbu　lent

characteristi’cs　of　lktu’k　FsolkS　fiow　i　n　the　riser　with　ub．　Figures　5．12　＆　5．13　respectively　show

tyl　Xceti　power　spectrel　density　（PS　D）　fu　nct）’ons　of　ti　me　series　data　of　opticeil　and　heat　transfer

flu（加ab’ons　obtEained　at　ub＝0．320　m・♂，輔吻e　pio制。n　l（㈱mb調e　diag㎜s．睦

can　be　seen　that　the　PSD　funcb’on　for　locEal　voiclage　fl　uctuation　seems　a　broad　spectra　with

many　sp’kes　in廿1e　vvk把range　of　frequency　while　heat　transfer　fiuctuation　deaHy　exhb硅s　a

conti　n　uous　power　spectra　with　an　exponentiat　decay　in　the　range　of　higher　frequency．　This

implies　that　the　bed　vok　lage　fiuctuates　in　the　random　or　stochastic　man　ner，　suggesting　much

complex　stnJctu　re　of　attractor．　（）n　the　oth’er　hand，　chaotic　characteristic　in　heat　transfer

fluctuation　can　be　expected　from　the　latter　PS　D　funcijon　（Fig．5．1　3）．

　　　Rgu【e　5．14　shows　typk）d　resu忙of　the　k：劇hmk）plots　of廿1e　oorrelation　integr剖，　Cの，

with　radius　of　hyper　sphere，　r，　at　various　em　beoding　dimensions　for　the　output　signat　of　the

optical　fiber　probe　at　ub　＝　O．320　m・s’i　and　G，　＝　35．61　kg・m2・s”．　For　the　scetiing　regions
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dominated　by　the　frac圃strudu　re　in　the　reconstruded　atセactor，　the　k）garith　m　ic　plots　of　C（tナィ

for　a　giVen　d　were　observed　to　be　correlated　by　a　sセ窃Ught　line．　The　correlatjon　dimerw’on　for

each　embeckiing　dimension　was　obtained　from　the　slopes　of　the　correlated　line．

Fu曲ermore，　as　am　be　seen　in　fig．5・15，　Dc　increases　wtth　i　n　the　smat1　range　of　d　and　levels

off　when　d　becomes　layge　enou　gh　to　captu　re　the，　system’s　complexity　（Takens；　1　981）．　Thus，

D，　for　the　range　of　d，　where　the　slope　in　C（r）一r　plots　becomes　independent　of　the　embedding

dimension，　is　hereafter　employed．

　　　臼gu陀s　5．16＆5．17　show　the　van’ations　of　the　oorrelation　dimension　and　Kolmogorov

entropy　of廿1e　oP圃and　heat　transfer　flu（加atjons　wi廿1　voidage　at　d’nferent　ub．　As　can　be

seen　in　Fig．5．16，　Dc　for　the　opticaj　signals　tend　to　dectrease　graduatly　with　decrease　i　n　the

vcidage，　suggesb’ng　the　formation　of　the　oontinuous　solld　phase，　as　described　above．　（）n

廿1e　o廿ler　hand，　Dc　for廿1e　heat　transfer　is　found　to　be　decreased　slヒ〕htiy　with　increase　in廿1e

voidage　within　the　present　experimentat　oondin’ons．　ln　addiv’on，　Dc　for　the　heat　transfer

fluctuation　at　ub　＝　O．320　m・s’i　becomes　large　in　comparison　with　that　at　ub　of　O．1　07　m・s’i　and

O．139　m・s－i．　From　these　results，　no　significant　change　in　the　chaotic　characteristics　of　local

fiow　structu　re　of　the　partbes　wlth　ub　um　be　seen　within　the　present　experimental　condhions．

On　the　other　hand，　the　complexity　of　th　e　local　heat　transfer　fiuctuation　becomes　higher　as

incteasing　ub　This　may　be　attn’nme　to　the　promotion　of　the　k）cal　turbulence　whicSi　ts

induced　from　the　formation　of　the　aggregates　of　the　nves　and　th　e　streak　of　the　fiuidta’ng

lktuid．　l　n　fig．5．17，　Kolmogorov　entropy　obtatned　in　the　present　operating　cond’nions　are

found　to　be　atways　positive，　which　suggests　the　chaotic　nature　in　the　both　fluctuations．　In

add面on，　the　entropy　of廿1e　OρtiCEal　fi　u（カ」ation　consberab㌧depends　on廿1e　vok＝bge　while　that

for　the　heat　fi　ux　seems　to　become　similar　values　in　the　rarige　between　4　to　5．5，　at　any　vatue

of　the　voiciage．　This　suggests　that　the　degree　of　freedem　in　the　motion　of　the　fiuidized

partides　van’es　matnly　with　bed　va’dage　wh　ile　no　sign’rfcant　effects　of　ub　and　vo’nage　on　the

chaotic　motion　of　waterLsolkl　slurry，　wh　ich　is　preclominant　for　the　local　heat　trmsfer，　are　seen

in　the　present　experimentail　oondin’ons．
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Fig．5．16　Comparison　of　variation　in　correlation　dimension　for　local　voidage

　　　　　　and　heat　transfer　fluctuations　as　a　function　of　voidage，　obtained

　　　　　　at　different　liquid　velocities
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5．4　ConcS　usion

　　The　local　bed　vok　iage　and　heat　transfer　fi　uctuations　were　simuttaneously　measured　in

廿1e　drcula哲ng　IkquiCFfiuiritzed　bed　riser　at　wkle　range　of　superfida　I　liqU’id　vel（x蹴y　and　drα」lating

sc”id　mass　flux　arrd　the　deterministi’c　chaos　analysis　was　adopted　to　charactertze　the　thor’r

chaotic　c2iaracteristjcs．　Within　the　present　experimental　condta’ons，　followings　are　darified：

1）　11me．averaged　heat　transfer　coefficient，　h，　in　CLFB　riser　is　higher　than　that　in　the

　　conventionat　batch　ljou’xtFfiuidtzed　bed．　H　igher　h　can　be　obtained　at　ub　appreciably

　　higher　than　the　term　inat　velocity，　where　the　nve　bed　exhibits　the　aggregatix／e

　　fluidtZation　sセヨ：te．

2）　L（x）dheat旋msfer　b　promoted　when　I（x）d　bed　vok：拾ge　decreases．　This　is　due　to

　　tt」rt：x」lent　convec極on　and　renewal　of廿le　surface　arて）und廿1e　heat　transfer　probe．

3）　The　power　spectrat　dens’tty　fu　nctjons　of　the　voiclage　and　heat　transfer　fl　uctuations　exhibit

　　a　continuous　power　spectru　m　with　exponentiEti　decay，　suggesting　that　the　hydrodynarnic

　　behavior　of　Ik】ukおokゴsystem　is　chaotk）．

4）　Correlation　dimension　for　loceti　bed　voiclage　fiuctuation　is　graduatly　decreases　with

　　decrease　in　bed　voiciage，　suggesting　the　formation　of　the　continuous　solld　phase　in　the

　　n’ser．　Corresponding　to　the　decrease　in　the　vok　iage，　time．averaged　heat　transfer　is

　　increased　wh　ich　is　due　to　the　promotion　of　locel　tu　rbu　lence　of　the　ltauid－solkl　flow　and

　　renewat　of　the　su　rface　in　the　thermat　boundary　layer　arou　nd　the　heat　transter　probe．
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Chapter　6

GENERAL　CONCLUSION

　　　This　thesis　is　a　summary　of　series　of　experimenta1　work　which　has　been　canied　out　in

an　effort　to　make　clear　the　flow　phenomena　of　solid　parl）’cles　fluidized　by　gas　and　liquid　at

van’盾浮刀@velocde’es　higher　than　the　term　inal　velocity　of　the　single　panicle，　ut．　Systematic

measurements　were　oonduCted　to　elucidate　flow　pa廿erns　of　sblid　particles　in廿le　risers　of

ciroulating　gas一　and　liquid－fluidized　bed　（CG　FB　and　CLFB，　respectively）　systems．

　　　　For　CGFB　system，　attention　has　oonsistently　been　focused　to　the　effects　of　the

system　variables　such　as　the　height　of　riser　and　the　initial　soiid　inventory　in　the　downcomer

on　steady－state　and　unsteady」state　axial　bed　denslty（ABD）disthbutions　in廿1e　hser．　The

effects　have　been　examined　measuring　disbibutions　of　static　pressures　along　the　height　of

risers　of　5．5　m　and　10　m　high　and　O．1　m　i．d．　and　of　downcomer　in　a　CGFB　system．　The

measurements　were　conducted　under　the　conditions　where　the　circulating　solid　mass　flux，

Gs，　was　controlled　independently　of　pressure　balance　between　the　riser　and　the

downcomer　bottoms　as　well　as　of　the　superficial　gas　veloc’rty　in　the　riser，　uge．　Within　the

present　experimental　oonditions，　the　followings　were　evidenced；

1）　When　dilute－phase　transport　or　dense－phase　transport　state　appears　in　the　riser，　ABD

　　　distributions　at　a　fixed　set　of　values　of　uge　and　Gs　are　generally　independent　of　the

　　　initial　solid　inventory，　Hsd，　in　the　downcomer．

2）　When　G，　takes　a　particular　value　at　a　given　gas　velocity，　S－shaped　ABD　distribution

　　　appears，　which　consists　of　a　dense　region　in　the　lower　part　and　a　dilute　region　in　the

　　　upper　part　of　the　riser．　At　such　a　condition，　the　height　’of　the　dense－region　depends

　　　appreciably　on　Hsd．　This　can　be　explained　by　an　analogy　to　existing　models　for　solid

　　　flow　in　the　freeboard　of　the　bubbling　fluidized　beds．

3）　The　dense－region　height　increases　with　the　riser　height　even　under　condhions　where

　　　all　the　other　operating　variables　are　the　same．　Thus，　an　apparent　overtap　of　the　ABD

　　　distribし面on　curves　for　risers　of　the　different　heights　is　obtainable　when廿1e　disbibution

　　　　　　　　　　　　　　　　　　　　　　　　　　　一“1一
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ourves　are　drawn　by　leveling　the　riser　exit．

Transin’onal　variations　are　observed　in　Pro　and　P，b．　The　variation　reflects　not　only

disbibutr’on　of　solid　panicles　iniv’ally　loaded　in　the　downcomer　to　the　riser　and　the

downoomer　but　aiso　slow　development　of　the　static　pressure　distn’btm’on　in　the

downcomer．

The　time　required　to　attain　a　steady　distn’bution　is　longer　for　Pdo　than　for　Pro　due　to　the

slow　gas　flow　from　the　riser　to　the　downcomer　and　the　induced　slow　gas　flow　in　the

upper　part　of　the　bed　in　the　downcomer　by　the　solid　pafu’cles　raining　down　from　the

cyclones　onto　the　bed．

No　appreciable　effect　of　the　variation　in　Pcb　is　found　on　steady－state　ABD　distributions．

Only　when　Gs　takes　a　pari）’cular　value　at　which　S－mshaped　ABD　distributions　appear　in

廿1e　riser，　pr◎and　correspondingly廿1e　ABD　distribu恒on》ary　w辻h　P（め．　At　such　a

cond詮ion，廿ie　steady」state　ABD　distribtm’on　depends　on　Hsd．

　　　A　further　analysis　of　fundamenta1　high－velocity　flow　pattems　of　solid　panicies　were

conducted　employing　a　CLFB　system　wth　a　4．0　m　high　and　O．049　m　i．d．　riser　to　simplify

the　above　oomplicated　and　unstable　flow　characteristics　in　CGFB　system．

Measurements　were　canied　out　at　various　Gs　and　the　superfictal　liquid　veloc’rty，　ub，　for　ABD

distn’bution　in　the　riser　as　well　as　fluctuations　in　the　local　bed　voidage　and　heat　transfer　rate

both　of　which　strongly　reflect　the　particle　flow　pattems　in　the　riser　and　the　followings　were

found：

1）　The　S－shaped　distributions，　which　are　frequently　observed　in　CGFB　risers，　are　never

　　　produced　in　the　CLFB　riser，　i．e．　ABD　distributions　are　always　uniform　along　the　CLFB

　　　riser．　Existing　flow　models　obtained　for　pardculate　or　packed　beds　are　inapplicable　to

　　　describe　the　relationship　between　the　liquid－solid　slip　velocity　and　the　bed　voidage　in

　　　CLFB．

2）　Even　a　bed　which　exhibits　pardculate　fluidization　state　at　ub　〈　ut　tends　to　form

　　　aggregates　in　CLFB　riser　at　ub　〉　ut，　where　the　fluidizing　liquid　eventually　passes

　　　through　the　bed　in　a　form　of　streak．

3）　Spatial　distribution　of　voidage　in　CLFB　riser　exhibits　frac　tal　structure．　Lower　fractal

　　　dimensions　are　obtained　at　ub　m　uch　higher　than　lh，　which　is　attributed　to　formation　of

　　　iarger　scale　aggregates．

4）　Time－averaged　heat　transfer　ooethcient，　h，　is　appreo’ably　higher　than　those　obtained　in

　　　batch　Iiquid　fluidized　beds．　Afur廿1er　increase　of　h　takes　place　as　ub　is　increased　over
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ut，　where　th　e　bed　exhibits　the　aggregative　fluidization　state．

The　power　spectrat　density　fu　nct）’ons　of　the　heat　transfer　fl　uctuations　exhibit　a

oontinuous　power　spectru　m　with　exponential　decay，　suggesting　that　the　state　of　liquid－

fluidization　at　ub＞しlt　is　chaotic．

Heat　transfer　coefficient　increases　with　increasing　ub　and　further　with　decreasing　lcx　）al

voidage　at　a　fixed　ub．　Higher　correlation　dimensions　for　the　local　heat　transfer

fiuduatbn　are　obtained　probably　due　to　turbulent　convection　and　renewal　of廿）e

surface　around　the　heat　transfer　probe．

　　　From　the　analysis，　it　becomes　c［ear　that　the　fluctuation　in　the　heat　transfer　inherentiy

oontatns　chaotic　natUre．　This　suggests　that　attainabie　flow　state　in廿1e　hser　is　complex　but

deterministic．　The　next　step　in　this　research　field　vvill　be　synchronization　of　the　predicted

fluctuation　by　using　a　neural　network　with　the　actual　fluctuation　among　different　operating

condin’ons　or　riser　scales　based　on　the　chaotic　statistics．
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Nomenclature

一
，
0

Cω

P

Pbo

P．

APma

APrd

r

s

t

t
s
tsamp

Ua

uge

Ub

U
p

　cross－sectjonat　area　of　riser

the　average　number　of　time　steps　necessary　for　an　iniv’al

　　　interpoint　distance　to　grow　beyond　a　specific　length

oorrelation　integral

oorrelation　dimension

fractal　dimension

embedding　dimension

mean　parb’cle　diameter

solid　mass　fiux　in　riser

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　tcircu［ating　solid　mass　flux

grav’itational　acceleration

height　in　downoomer

static　bed　height　initially　loaded　in　downcomer

heat　transfer　ooerncient

Kolmogorov　entropy

a　specific　maximum　distance

total　number　of　POints　on　recons加ded　a廿rador　used

　　for　oalculation　of　a　correlation　integral

static　pressure

static　pressure　at　downcomer　bottom

static　pressure　at　riser　bottom

pressure　drop　through　riser

pressure　difference　between　riser

　　and　downcomer　bottoms

radius　of　a　given　hypersphere　in廿）e　phase　space

mesh　scale

time　elapsed　after　the　onset　of　solid　circulation

ti　m　e　required　to　attain　steady－state　for　C　FB　system

time　for　water－solid　slurry　sampling

rate　of　aeration　to　downcomer　bottom

superficial　gas　velocity

superficial　liquid　velocity

particle　velocity

　　　　　　　　　　　　　　　　－114一

響一

［　M2］

　　　　　　［．一］

　　　　　　卜］

　　　　　　日

　　　　　　卜］

　　　　　　卜］

　　　　　［m］

［　kg・m’2．s”　］

［　kg・m－2．s’i］

　　　［M．s’2］

　　　　　［m］

　　　　　［呵

［W・m’2．s’i　］

［　bits／step　］

　　　　　日

卜］

［Pa］

［Pa］

［Pa］

［Pa］

　　［　Pa］

　　　臼

　　　［一］

　　　［s］

　　　［s］

　　　［s］

［　m3．s’i　］

［　M．s’i　］

（　M．s’i　］

［　M．s’i　］
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u
t
V
s
Ws

↓
X

　α
X
　
Z

　
　
⇒

ε
亀
釧
K
μ
ρ
飾
儀
τ

liquid－solid　slip　velocity

terminal　velocity　of　a　single　panicle

volume　of　sampled　water－solid　slurry

weight　of　sampled　water－solid　slurry

phase　space　vectors　reconstructed　from　time－series

time　series

height　from　distributor　in　riser

mean　bed　voidage

voidage　of　sampled　water－soiid　slurry

Heaviside　function

tortuosity　of　fiuid

liquid　visoosity

liquid　density

particle　density

solid　concentration　of　water－solid　slurry

time　interval

［　M．s’i　］

［M．s’i］

［m3］

　［kg］

卜】

　　　卜］

　　［m］

　　　卜］

　　　日

　　　臼

　　　日

［　Pa・s　］

［　kg・m“　］

［　kg・m“g

［kg・mB］

　　　［s］
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